

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Carrasco, Bruno; Choi, Seung Mo

Working Paper An Assessment of Cross-country Fiscal Consolidation

ERD Working Paper Series, No. 79

Provided in Cooperation with: Asian Development Bank (ADB), Manila

Suggested Citation: Carrasco, Bruno; Choi, Seung Mo (2006) : An Assessment of Cross-country Fiscal Consolidation, ERD Working Paper Series, No. 79, Asian Development Bank (ADB), Manila, https://hdl.handle.net/11540/1873

This Version is available at: https://hdl.handle.net/10419/109276

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/3.0/igo

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



ERD Working Paper ECONOMICS AND RESEARCH DEPARTMENT SERIES No. 70

An Assessment of Cross-country Fiscal Consolidation

Bruno Carrasco and Seung Mo Choi

February 2006

Asian Development Bank

ERD Working Paper No. 79

AN ASSESSMENT OF CROSS-COUNTRY FISCAL CONSOLIDATION

BRUNO CARRASCO AND SEUNG MO CHOI

February 2006

Bruno Carrasco is Principal Financial Economist in the Governance, Finance and Trade Division, South Asia Department, Asian Development Bank (ADB). Seung Mo Choi is a Ph.D. student in the Department of Economics at the University of Chicago. The authors thank Alberto Alesina and participants of the Seminar held on 19 August 2004 at ADB and of the Public Finance and Expenditure Management Course held on 8 November 2005 at the ADB Institute for valuable comments. Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/economics

©2006 by Asian Development Bank February 2006 ISSN 1655-5252

The views expressed in this paper are those of the author(s) and do not necessarily reflect the views or policies of the Asian Development Bank.

FOREWORD

The ERD Working Paper Series is a forum for ongoing and recently completed research and policy studies undertaken in the Asian Development Bank or on its behalf. The Series is a quick-disseminating, informal publication meant to stimulate discussion and elicit feedback. Papers published under this Series could subsequently be revised for publication as articles in professional journals or chapters in books.

CONTENTS

Abstra	ct		vii
I.	INTRO	DUCTION	1
II.	LITERA	TURE REVIEW	2
III.	FRAME	WORK AND METHODOLOGY	3
	A. B. C. D.	Initial Conditions Composition of Adjustment Initial Conditions vs. Composition of Adjustment Methodology	4 5 5 5
IV.	FISCAL	ADJUSTMENT	8
	A. B. C. D. E. F.	Primary Deficit and Government Debt Primary Expenditure and Total Revenue Components of Expenditure and Revenue Capital Expenditure in EMEs Expenditure on Interests Removing Capital Expenditure	8 11 13 17 17 20
۷.	MACRO	ECONOMIC ADJUSTMENT	20
	A. B. C. D.	Why Fiscal Adjustment is Not Necessarily Contractionary GDP Growth Rate and Unemployment Rate Role of Initial Fiscal Conditions on Macro Variables Probit Analysis	20 24 36 36
VI.	CONCLU	JSION	39
	Appeno Referer	dix nces	41 47

ABSTRACT

This paper analyzes fiscal adjustment by reviewing the cross-country experience of the Organization for Economic Cooperation and Development (OECD) or mature economies, and the emerging market economies during 1970-2002. We find that initial conditions, composition, and timing of adjustment matter for successful fiscal adjustment (SFA). First, countries with higher primary deficits and larger government debts are more likely to pursue SFA. Second, in mature economies, SFA is driven by expenditure compression measures (largely cutbacks in subsidies) rather than by revenue augmentation measures. On the other hand, in emerging market economies, while expenditure compression measures (in particular, a decrease in capital expenditure) drive SFA, revenue augmentation measures also play an important role. Third, synchronizing fiscal adjustment to an expansionary phase of the business cycle increases the likelihood of its success. A probit analysis tends to support these findings. Finally, we review whether fiscal adjustment can indeed be expansionary. The findings point to a weak inclination toward noncontractionary outcomes, although further research is required to explain what drives this non-Keynesian result.

I. INTRODUCTION

Fiscal imbalance and measures to address this are once again an issue of heightened concern for policymakers. In mature economies (MEs) such as of the Organisation for Economic Co-operation and Development (OECD), fiscal deficits reflect a combination of greater global imbalances rooted in structural constraints and linked to a constellation of policy objectives ranging from tax-cutting measures (United States), to bank restructuring imperatives and measures to reflate the economy (Japan), to a tax and spend culture in support of a social model of development (European Union [(EU]). In emerging market economies (EMEs), fiscal deficits generally reflect large demands on public expenditure by way of employment absorption and infrastructure development combined with limited revenue generating capacity.

Governments across the world struggle with measures to reduce fiscal deficits, recognizing the long-term benefits of fiscal consolidation together with the complexities and costs associated with short-term fiscal adjustment. Large fiscal deficits generally reduce national saving and capital formation, leading to adverse consequences including reduced economic growth, lowered real incomes, and even increased risks of financial crisis. Against this backdrop, the underlying consensus has tended to favor the adoption of some form of fiscal balance over the cycle (see, for example, OECD 2003).

This paper focuses on the analysis of fiscal adjustment¹ or the process by which governments attempt to address fiscal imbalance, by reviewing the cross-country experience of both the OECD countries or MEs, and emerging market economies over the 1970–2002 period. We attempt to address key questions that policymakers are typically confronted with: (i) When does fiscal adjustment lead to successful fiscal consolidation?² (ii) What drives *expansionary* fiscal adjustment? (iii) Do initial conditions matter for fiscal adjustment and fiscal consolidation? (iv) Is the composition of fiscal adjustment relevant? (v) Are there differences in characteristics between successful fiscal adjustment in emerging market economies and MEs?

The paper extends the results of Ardagna (2004) and Alesina and Ardagna (1998) by including a wider sample estimation that covers EMEs and provides further evidence of standard results. However, it also provides new insights into commonalities and differences between ME and emerging market economy subsamples that may provide valuable lessons to policymakers (who may otherwise have generalized results from MEs).

Our results support the view that while expenditure compression measures drive successful fiscal adjustment in MEs, in emerging market economies the combination of *revenue augmentation*

¹ Fiscal adjustment in this paper is defined as the active process of compressing expenditure and/or augmenting revenue with the objective of improving the fiscal balance.

² While successful fiscal adjustment is defined more formally in Section III, for the time being it can be understood as a fiscal adjustment process that leads to an improvement in the fiscal balance.

measures and expenditure compression drives this process. However, a closer look at the adjustment process supports the view that the *initial conditions* also matter in explaining successful fiscal adjustment.

According to our results, on average, MEs have higher revenue yields compared to EMEs, suggesting little room for introducing further revenue augmentation measures to correct fiscal imbalance. In parallel, our results indicate that, on average, expenditure to gross domestic product (GDP) ratios are higher in MEs than EMEs, providing further room in MEs to introduce expenditure cutbacks without undermining the provision of essential public goods and services. On a more worrying note, our results suggest that in those EMEs that have managed to successfully reduce their fiscal deficits, one half of the decrease in primary expenditure is on average due to a decrease in capital spending.

Turning to the question of when is the policymaker more likely to succeed in fiscal consolidation, our results suggest that coinciding with the expansionary phase in the economy, *the timing of the introduction of fiscal adjustment matters,* and supports, synchronizing fiscal adjustment measures. Furthermore, on average, EMEs have higher debts and attendant debt servicing costs—or deadweight losses—than MEs and hence the opportunity cost to the policymaker of not succeeding in fiscal consolidation is far greater than in economies with lower debts.

Our findings also point to a common fiscal adjustment process that leads to non-Keynesian results, that is, *successful fiscal adjustment does appear to have an impact on the economy—it decreases the unemployment rate steadily*. This is consistent with the literature (see Alesina, Perotti, and Tavares 1998) and attributed to the supply-side effects from fiscal consolidation processes, which, if deemed credible by economic agents, may lead to expectations of higher household disposable incomes (through lower taxes) and higher profits for businesses (from lower taxes and interest costs), resulting in an increase in aggregate demand in the economy. Finally, our results point to the importance of sustainability of the adjustment process. Adjustment processes that extend beyond one year are likely to be more successful in reducing primary deficits. Our results are in line with Obstfeld (1998) who argues that the effects of fiscal adjustment will depend on initial conditions, particularly on whether fiscal policy is on a sustainable course.

Section II of this paper presents a review of the literature. Section III presents a heuristic framework that brings together initial conditions and composition of adjustment considerations and describes the methodology and definitions. Section IV reviews results from the sample of countries focusing on the fiscal adjustment process. Section V reviews results focusing on macroeconomic adjustment and explains why fiscal adjustment may be expansionary. Section VI concludes.

II. LITERATURE REVIEW

The applied literature on fiscal adjustment was significantly influenced by Alesina and Perotti (1997) who studied how the composition of adjustment influences both the likelihood of success and the macroeconomic consequences across OECD countries. Their findings suggested that composition does matter. In particular, fiscal adjustments resulting from spending cuts in transfers and the government wage bill (type 1) have both a better chance of success and are likely to lead to expansionary effects as compared to adjustments that rely primarily on tax increases and cuts in public investment (type 2).

To explain the greater likelihood of success, the authors referred to type 1 adjustment as inducing a more lasting consolidation of the budget as they tackle two key items, government wages and welfare programs, which have the strongest tendency to automatically increase. To explain the macroeconomic consequences, Alesina and Perotti (1999) refocused the arguments away from conventional wisdom that referred to how credibility and wealth effects of adjustment would affect consumption and emphasized the effects of fiscal policy on unit labor costs and competitiveness. Indeed their results suggest that unit labor costs may be more empirically relevant for consumption than the wealth effects and credibility channel.

Alesina, Perotti, and Tavares (1998) further developed the importance of unit labor costs in explaining the supply-side effects from fiscal consolidation processes. According to them, if the announced fiscal policy effects on unit labor costs are deemed credible by economic agents, this may lead to expectations of higher household disposable incomes (through lower taxes) and higher profits for businesses (from lower interest costs), resulting in an increase in aggregate demand in the economy. These results contributed to a literature that was characterized by non-Keynesian results, that is, that fiscal contractions could indeed be expansionary as they signaled a permanent and decisive change in the stance of fiscal policy (see for example, Giavazzi and Pagano 1996).

More recently, Ardagna (2004) has posited two possible explanations of successful fiscal adjustment: the "expectation view" and "labor market view." The former refers to the impact of the fiscal stance on the economy through its influence on agents' expectations (i.e., what type of expenditure contraction and/or revenue augmentation measures are deemed credible) and the latter focuses on the effect of the composition of adjustment on the economy (i.e., do reductions in growth of wages and salaries or pensions lead to more favorable supply-side effects on the economy). Blanchard (1990a) and Bertola and Drazen (1993) are considered to fall under the former category. In their models, initial conditions may influence expectations formation and, in particular, changes in expectations in economies "close to the edge" could prove expansionary. Alesina and Perotti (1995 and 1997) are included under the latter, stressing that expenditure reduction programs succeed because they induce higher growth. Bertola and Drazen (1993) attempt to reconcile the differing results by suggesting that while in a static model the fiscal policy measures may be contractionary, it could well be that in a dynamic, forward-looking model, it induces sufficiently strong expectations of future policy changes that may lead to expansionary effects.

III. FRAMEWORK AND METHODOLOGY

In our view, any stylized model should (i) attempt to answer the question of what is it that makes a fiscal adjustment process turn out to be successful and (ii) integrate these explanations into a unified structure. An alternative explanation to Ardagna (2004) is described below.

To begin, we postulate that the success of fiscal adjustment will critically depend on whether economic agents perceive the *adjustment* to be credible or not. We consider two polar cases across the spectrum.

If fiscal adjustment is deemed *credible*, forward-looking, rational agents—firms and households—will adjust behavior following the fiscal announcement or shortly thereafter. Firms

would anticipate that the announced fiscal measures would result in a combination of *future* lower interest rates, perhaps lower corporate taxes and even a pick-up in aggregate demand and, *ceteris paribus*, would react by increasing capital expenditure *today*. Similarly, households would anticipate *future* higher disposable income from lower income taxes as public spending is cut and expand private consumption *early on*.

The key is that the credibility of the announcement generates an *endogenous adjustment process* in agents' behavior. The resulting expansionary effect may, in turn, lead to a *virtuous cycle* where supply side factors may result in an increase in economic activity, which may stimulate demand and sustain higher tax revenues. Resulting higher economic growth would reduce the political costs to the policymaker of *sustaining* the fiscal measures in place, hence ensuring a degree of *persistence* of the fiscal consolidation process and thereby validating the rational expectations equilibrium.

The point to highlight is the interaction between credibility of the announced actions and the resulting behavior of economic agents. It is the adjustment by private agents triggered by the credibility of the announcement that accounts for the resulting expansionary impact on the economy and explains the non-Keynesian results. The fiscal adjustment is therefore considered *successful*.

If fiscal adjustment is *not credible*, forward-looking rational agents are not likely to react to changes in taxation policy and or expenditure policies and hence there is no endogenous adjustment process that kicks in. As a result, there is no expansionary effect and the tight fiscal stance is too costly for the policymaker to maintain. The fiscal measures prove to be unsustainable thereby validating the rational expectations equilibrium. The fiscal adjustment is therefore considered *not successful*.

In the analysis above, the credibility of the policymaker drives the results. An important issue therefore is how to assess credibility. While it is not possible to measure credibility, we can *ex post* infer from the size and timing of the behavior of (a pick-up in) consumption expenditure and private investment whether the fiscal measures were deemed credible. However, it would be important to be able to control for other influences on these variables.

While credibility drives the model, on the other hand, *initial conditions* and *composition of adjustment* do matter.

A. Initial Conditions

In the presence of significant fiscal imbalance at the time of the fiscal reforms announcement (i.e., large debt to GDP), there may be nonlinearities associated to the supply-side adjustment path that may lead to expansionary effects on the economy. If the initial conditions are particularly bad, this expansionary effect from fiscal correction could conceivably happen with less than full credibility on the part of economic agents. In this same spirit, and based on political economy, the political orientation of the government in power may partly influence the degree of credibility of the announced fiscal reform measures. This variable could have an even greater impact on the expectations with regard to the persistence of the fiscal tightening measures. This would be in a manner similar to the conservative inflation-hawk central bank governor who is more likely to

lean against the wind for a longer time for the sake of entrenching his inflation-fighting credentials on the market.

B. Composition of Adjustment

The composition of adjustment is important as it influences the particular adjustment path of the economy through its effect on economic agents' behavior. For example, increasing taxes over and above the increase in spending is not likely to be equivalent to decreasing spending in excess of the decrease in taxation. What is critical to bear in mind is that in the end it is the reaction by economic agents that is key in determining whether a fiscal adjustment is sustainable.

C. Initial Conditions vs. Composition of Adjustment

An interesting question is which of the two dominates, initial conditions or composition of adjustment? Prior beliefs would suggest that *ceteris paribus*, composition of fiscal adjustment matters more than initial conditions to the extent that it directly influences credibility. Consider, for example, untouchable budget items such as transfers, wages, and public employment. Cutbacks in these items are clearer signals of the government's intent on "biting the bullet" than, for instance, politically less painful measures such as cutbacks in investment spending, reductions in the operations and maintenance budget, or even increases in taxation.

Accordingly, the pay-off matrix in Box 1 could summarize the analysis above.

Box 1 Fiscal Adjustment Matrix Initial Conditions							
COMPOSITION	LOW DEBT	HIGH DEBT					
Expenditure Contraction	М -	S					
	NC	Μ.					

D. Methodology

Our main interest in pursuing this research is to understand what happens during and after a "successful" (and/or "sustained") fiscal adjustment: For example, which components in the government expenditure (or revenue) are driving "successful" fiscal adjustment on average? Is "successful" (or "not successful") fiscal adjustment expansionary or contractionary in the short run? In addition, we also try to reconcile the results with the heuristics presented above. As a first step, it is useful to establish a metric to be able to compare fiscal behavior across countries and time. Accordingly, the following three concepts are defined below: (i) fiscal adjustment, (ii) "successful" fiscal adjustment, and (iii) "sustained" fiscal adjustment.

A period of **FISCAL ADJUSTMENT** is the year in which the primary balance (which, by definition, excludes the expenditure on interests) improves by at least 1.5% of GDP. That is, the year t is a period of fiscal adjustment if and only if

$$PD_{t} < PD_{t-1} - 1.5\%$$
,

where PD_t is the cyclically adjusted³ primary deficit as a percentage of GDP in year t. This definition is adapted from Alesina, Perotti, and Tavares (1998).⁴

(ii) A fiscal adjustment in year t is SUCCESSFUL (and referred to as SFA) if one of the following two conditions applies: either, (i) in the 3 years after the adjustment year, the ratio of the primary deficit to GDP is on average at least 2% below its level in the initial year, that is,

$$(PD_{t+1} + PD_{t+2} + PD_{t+3}) / 3 < PD_{t} - 2\%,$$

or; (ii) 3 years after the year of the adjustment, the debt-to-GDP ratio is at least 5% below its level in the adjustment year, that is,

Debt
$$_{t+3}$$
 < Debt $_{t}$ - 5%.

We define a fiscal adjustment to be "not successful" (and referred to as NSFA) if none of the above conditions apply.⁵ This criterion is used by both of Alesina, Perotti, and Tavares (1998) and Alesina and Ardagna (1998).

Appendix Table 1 lists all 455 episodes of fiscal adjustment (110 SFA and 345 NSFA), identified by these definitions based on the "cyclically adjusted" primary deficits. As observed in the table, some specific countries may have more episodes of fiscal adjustment than others. Although this may raise considerations of robustness of the results, we treat all of those episodes in the same way in order not to lose any information that each case may reveal.

(iii) A fiscal adjustment is SUSTAINED if the fiscal adjustment (in our definition) continues in the following year. That is, the year t is a period of "sustained" fiscal adjustment if and only if

$$PD_{t} < PD_{t-1} - 1.5\%$$
, and
 $PD_{t+1} < PD_{t} - 1.5\%$.

³ For a description of the data including the cyclical adjustment procedure, see Appendix.

⁴ Alesina and Ardagna (1998) also use a similar but stronger definition: They consider a fiscal adjustment if in any year there is an improvement on the primary balance of at least 2%, or if in two consecutive years there is an improvement of at least 1.5% each.

⁵ If both "primary deficit data of the next 3 years" and "debt data after 3 years" are missing, a fiscal adjustment is considered neither "successful" nor "not successful."

By definition, if the year t is a period of "sustained" fiscal adjustment, then the year t+1 is a period of fiscal adjustment. Since we want to count two consecutive years of fiscal adjustment as one "sustained" episode, we do not count the year t+1 as either "sustained" or "not sustained" if the year t is the year of sustained fiscal adjustment to avoid double counting.

From the definitions, a "sustained" fiscal adjustment is more likely to be "successful" because it implies a subsequent strong improvement in the primary balance in the following year, and thereby decreases the three-year average of primary deficits used in the definition of "successful" fiscal adjustment. Appendix Table 2 confirms this conjecture. Out of 84 sustained episodes of fiscal adjustment, 52 episodes turn out to be successful, so the (sample) probability of success for sustained fiscal adjustment is about 62%. On the other hand, out of 295 "not sustained" episodes, only 35 were successful, or a probability of success of about 12%.

As is pointed out by Alesina and Ardagna (1998), this definition of fiscal adjustment is relatively strong. This strong definition enables us to focus on the adjustment episodes resulting from proactive, fiscal consolidation-oriented policy engagements.

As a second step, we calculate how the descriptive statistics of the variables of interest evolve over time, that is in years t-1, t, t+1, t+2, etc., where "successful" (or "not successful") fiscal adjustment is initiated in year t. For example, by comparing the mean values of the GDP growth rate⁶ in years t-1 and t+2, we can figure out whether it is increasing and hence fiscal adjustment is expansionary or contractionary, on average, during the fiscal reform process.

How the variables differ when a "successful" fiscal adjustment is achieved can be (at least partly) understood by considering how the mean values (and other statistics) change over time. For example, the average GDP growth rate across all countries (say, in Group 1) during the chosen time frame is 3%, and if the average GDP growth rate in case of SFA has increased from 3% in t-1 to 5% in t+2, then it would be plausible (although not certain) to conclude that economies tend to be better off following SFA.

Also, to understand how SFA differs from NSFA, we can compare the descriptive statistics in different years around SFA and NSFA outcomes. If the average GDP growth rates are the same in t-1 but "successful" cases show a higher average growth rate in t+2 and later, then we infer that a SFA is accompanied by a higher GDP growth rate.

This methodology, however, raises an econometrical issue; it is not ideal to compare SFA and NSFA episodes to understand what results from SFA episodes against what *would* have resulted if they were NSFA ones because NSFA episodes are not the counterfactuals themselves. Since we do not control for other variables (e.g., the monetary policy stance or whether there was an external shock) we are not able to assume *ceteris paribus* in comparing the results. This problem exists in similar studies, and we do note this caveat when we interpret our results.

⁶ In the actual analysis, we consider the real GDP growth net of its "trend" (calculated by Hodrick-Prescott filter) as a proxy for the business cycle.

IV. FISCAL ADJUSTMENT

In this section, we analyze the dynamics of key *fiscal variables* during SFA versus NSFA.⁷

A. Primary Deficit and Government Debt

To begin, we start by looking at how the level of primary deficits and government debts (as a percent of GDP), change over time during episodes of fiscal adjustment. By definition, it is straightforward to see that the improvements in primary deficits and government debts will be larger in case of SFA rather than NSFA episodes, and in case of "successful and sustained" (which are more painful to the policymaker) rather than "successful and not sustained" episodes. Such predictions are empirically observed in Table 1 (on primary deficit) and Table 2 (on government debt). We notice the following results in these tables:

- (i) Sample Averages: On average, Group 1⁸ (high-income OECD economies) has the lowest level of primary deficit (-0.4% of GDP⁹) and debt (35.9% of GDP). Group 3 (middle-income economies) has an average level of primary deficit (0.4% of GDP) and debt (43.6%). According to our sample, on average, the levels of primary deficit and debt are not necessarily related to the income level. At the same time, OECD economies seem to have relatively greater fiscal discipline compared to non-OECD economies.
- (ii) Initial Conditions: The levels of primary deficit and government debt at t-1 or period prior to the fiscal adjustment are much higher in SFA episodes than in NSFA ones across the two income groups. (This tends to support the view that given worse initial conditions, governments are generally under greater pressure and have greater incentives to be serious about fiscal adjustment, thereby increasing the likelihood of SFA).
- (iii) Magnitude of Improvements: The magnitude of improvements in both primary deficit and debt from t-1 to t+2 are, as predicted, much greater in SFA episodes than in NSFA episodes for all two income groups. (Primary deficits: -6.6 percentage points [pp] versus -1.9pp in Group 1, and -7.6pp versus -2.2pp in Group 3. Debts: -3.0pp versus 3.3pp in Group 1, -6.0pp versus 9.5pp in Group 3.¹⁰ Also, "successful and sustained" episodes always show greater improvements in primary deficits, on average, than "successful and not sustained" ones.¹¹

⁷ In this paper, for the purpose of easy reference, we limit the presentation to the results under Groups 1 and 3. Interested readers who would like to refer to results of Groups 2 and 4 are asked to contact the authors.

⁸ For the breakdown of the groups see Section A under Appendix.

⁹ Throughout the paper a negative primary deficit corresponds to a primary surplus.

 $^{^{\}rm 10}$ Group 3 shows an extremely large drop in debt, caused by a few episodes.

¹¹ Such a trend is not very clearly observed in government debt. However, for the first two income groups, the sample size is too small to conclude anything, and for Group 3, the movements of medians and maxima suggest that sample means do not seem to be representative enough.

TABLE 1									
MOVEMENT	0F	CYCLICALLY	Adjusted	P RIMARY	DEFICIT	(AS	PERCENT	0F	GDP)

ITEM		S	UCCESSF	UL			NOT SUCC	ESSFUL	
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Group 1:	Mean	2.6	-0.3	-2.5	-4.0	1.1	-1.8	-1.2	-0.8
High-income	(Diff w/ <i>t-1</i>)		-2.9	-5.1	-6.6		-3.0	-2.3	-1.9
OECD	Median	1.3	-2.0	-3.6	-3.7	1.0	-1.3	-0.9	-0.5
Economies	Min	-5.2	-7.2	-10.2	-10.2	-7.3	-10.2	-10.4	-8.7
[-0.4]	Max	15.7	14.1	7.4	1.4	14.3	6.3	8.1	15.7
	SD	5.3	5.1	4.1	3.5	3.2	3.0	3.3	3.7
	#		25 /	/ 1 / 24			70 /	′ 12 / 58	
Group 3:	Mean	5.9	0.0	-1.6	-1.7	2.4	-2.3	0.1	0.2
Middle-	(Diff w/ <i>t-1</i>)		-5.9	-7.5	-7.6		-4.7	-2.2	-2.2
income	Median	5.2	-0.1	-1.3	-1.5	1.7	-2.0	0.1	-0.1
Economies	Min	-9.2	-14.9	-21.6	-21.8	-17.5	-21.6	-21.8	-18.8
[0.4]	Max	24.2	13.0	12.5	17.3	24.0	13.8	17.7	14.8
	SD	7.0	5.5	6.0	6.5	5.3	4.9	5.4	5.5
	#		54 /	/ 5 / 49			159	/ 42 / 11	7

"Successful" versus "Not Successful"

"Successful and Sustained" versus "Successful and Not Sustained"

ITEM		SUC	CESSFUL	AND SUS	TAINED	SUCCESSFUL AND NOT SUSTAINED				
		t-1	t	t+1	t+2	t-1	t	t+1	t+2	
Group 1: High-income OECD Economies	Mean (Diff w/ t-1) Median Min Max SD #	3.2 1.3 -2.6 15.7 5.2	0.7 - 2.5 -0.4 -6.4 14.1 5.6 13	-2.4 -5.6 -3.6 -10.2 7.4 5.1 / 0 / 13	-4.0 -7.2 -3.7 -10.1 1.4 3.8	0.9 1.1 -4.3 7.7 4.7	-2.5 -3.4 -2.8 -6.9 1.7 3.4 5 /	-2.2 -3.2 -0.7 -6.7 0.8 3.3 1 / 4	-3.7 -4.6 -3.6 -7.8 -0.5 2.8	
Group 3: Middle-income Economies	Mean (Diff w/ t-1) Median Min Max SD #	8.8 6.7 -0.7 20.8 6.1	2.6 -6.2 2.4 -9.2 13.0 5.5 23	-2.5 -11.3 -1.5 -16.3 10.0 5.8 / 0 / 23	-1.5 -10.3 -1.0 -21.6 11.4 6.8	5.4 4.1 -1.6 24.2 6.9	-0.6 -6.0 -2.1 -6.2 8.7 4.3 19 /	1.2 -4.2 -0.4 -4.5 12.5 4.5 4.5 4 / 15	0.6 -4.8 -1.2 -6.9 17.3 5.2	

Note: Other tables in this paper follow a similar structure to this table. Consider "successful" fiscal adjustments under the top left half of Table 1. We consider two groups of economies: Group 1, with a cyclically adjusted primary deficit (as % of GDP) sample mean of -0.4%, and Group 3 with 0.4%. (These numbers are put in "[]" in the far left column with the names of the groups. The sample mean is calculated by taking a simple average over all available observations for this income group.)

For each of the two income groups, the table shows the sample means of the variable for the years from *t*-1 to *t*+2, and their differences *continued*.

		ΤΑΒ	BLE 2					
MOVEMENT	OF	GOVERNMENT	Debt	(AS	PERCENT	0F	GDP))

"Successful" versus "Not Successful"

ITEM			SUC	CESSFUL			N01	SUCCESSE	UL	
		t-1	t	t+1	t+2	t-1	t	t+1	t+2	
Group 1: High-income OECD Economies [35.9]	Mean (Diff w/ t-1) Median Min Max SD #	46.7 45.0 6.0 80.1 23.4	46.2 - 0.4 45.4 6.4 85.6 23.6 16	45.3 -1.4 44.7 6.3 93.8 25.4 / 6 / 10	43.7 -3.0 39.5 6.3 109.8 27.3	33.3 26.9 0.4 109.4 28.2	34.0 0.7 25.2 0.0 115.6 30.0 29	36.1 2.8 27.7 0.0 113.7 33.2 / 16 / 13	36.7 3.3 25.1 0.0 117.9 34.9	
Group 3: Middle-income Economies [43.6]	Mean (Diff w/ t-1) Median Min Max SD #	56.1 40.5 3.0 190.1 36.4	54.7 -1.4 43.7 2.6 152.4 31.8 51	51.1 -5.0 41.7 1.6 134.7 31.4 / 21 / 30	50.1 -6.0 43.7 2.6 152.4 33.3	41.3 33.3 0.0 243.8 35.9	40.6 -0.7 35.2 0.0 229.0 35.8 72	45.0 3.7 35.9 0.0 332.1 46.0 / 45 / 27	50.8 9.5 40.1 0.0 401.0 56.7	

"Successful and Sustained" versus "Successful and Not Sustained"

ITEM		SUC	CESSFUL	AND SUS	TAINED	SUC	CESSFUL	AND NOT SU	JSTAINED	
		t-1	t	t+1	t+2	t-1	t	t+1	t+2	
Group 1: High-income OECD Economies	Mean (Diff w/ t-1) Median Min Max SD #	47.5 38.9 6.0 80.1 26.9	47.4 - 0.2 44.3 6.4 85.6 28.0 7 /	46.1 -1.4 46.4 6.7 93.8 30.0 3 / 4	46.5 -1.0 42.9 6.3 109.8 35.2	49.3 49.5 28.0 70.4 17.6	50.8 1.4 51.7 30.4 69.3 17.3 4 /	51.9 2.5 50.7 29.9 76.1 21.8 1 / 3	47.8 -1.6 47.9 27.7 67.6 20.1	
Group 3: Middle-income Economies	Mean (Diff w/ t-1) Median Min Max SD #	49.9 39.2 3.0 112.1 29.5	49.5 - 0.4 43.7 2.6 108.7 26.8 19	47.3 -2.6 39.3 1.6 99.8 29.0 / 9 / 10	47.8 -2.1 39.8 2.6 109.8 30.3	55.8 38.45 11.3 134.7 36.9	56.8 1.0 42.6 13.0 152.4 37.3 20	51.0 -4.8 43.1 9.9 134.7 33.8 7 7 / 13	50.1 -5.8 42.6 9.3 152.4 36.6	

against the initial value of sample mean in year t-1 (which we call the "initial condition"). For example, Group 1's primary deficits in case of successful adjustments have the sample means of 2.6%, -0.3%, -2.5%, and -4.0%, in t-1, t, t+1, and t+2, respectively. The differences against the initial conditions are -2.9%, -5.1%, and -6.6% for t, t+1, and t+2, respectively. The table also shows the medians, minimums (Min), maxima (Max), and standard deviations (SD). The first number (out of three) in the cell denoted by "#" implies the total number of episodes. The second and third numbers are the numbers of episodes in which the observed values in year t+2 are higher and lower than those in year t-1 (i.e., "initial conditions"), respectively. For example, there are 25 episodes of successful fiscal adjustments for Group 1. For 24 episodes out of these 25, the primary deficit had dropped from t-1to t+2, while it increased in only one episode.

B. Primary Expenditure and Total Revenue

A key question facing the policymaker is what type of reform measure is more likely to contribute to SFA on average—decreasing primary expenditure or increasing total revenue? Furthermore, how are the movements of these two fiscal variables different between SFA and NSFA episodes?

Alesina, Perotti, and Tavares (1998) and Alesina and Ardagna (1998) analyzed the OECD fiscal data to answer these questions. Their results can be summarized as follows. For OECD economies, a larger part of improvement on primary deficit is, on average, due to a decrease in *primary expenditure* (which is total expenditure net of interest payments) rather than an increase in *total revenue*. At the same time, SFA and NSFA episodes are more clearly distinguished by the magnitude of the decrease in *primary expenditure* than that of the increase in *total revenue*.

To confirm their findings and extend the analysis to EMEs, Table 3 reports the movements of these two fiscal variables during episodes of fiscal adjustment. The result supports the arguments of previous studies on OECD economies, although—as will be made apparent—EMEs seem to demand special attention:

- Sample Averages: The sample average of primary expenditure and total revenue (as % of GDP) over the whole sample is higher for Group 1 (about 30-31%) than for Group 3 (about 25-26%). That is, government spending and total revenue account for a smaller share of GDP in middle-income economies than in high-income economies.
- (ii) Initial Conditions: The average levels of primary expenditure and total revenue at t-1 are higher in SFA episodes than in NSFA ones in Group 1 while in Group 3 the results hold for primary expenditure although the two total revenue figures are the same. This is probably because when the government share of GDP is already larger, governments are under greater pressure and hence have greater incentives to decrease expenditure.
- (iii) Magnitude of Adjustment—Reason for Successful Episodes: For Group 1, in the case of SFA episodes, the primary expenditure has decreased on average by 4.4pp of GDP over three years, while the total revenue has increased by only 1.6pp. This implies that for Group 1, expenditure contraction contributes about 2.7 times more to successful fiscal adjustment than what revenue augmentation does, which confirms the results of previous studies. On the other hand, for Group 3, expenditure contraction (4.0pp reduction) contributes by 1.8 times more than what revenue augmentation (2.2pp increase) does. The bottom line is that for MEs (Group 1), expenditure contraction relative to revenue augmentation is more effective to reduce fiscal imbalance than for EMEs (Group 3). A possible explanation is that the revenue as share of GDP is on average higher in MEs, so there are no alternative means other than to reduce expenditure.

ITEM			SUCC	ESSFUL			NOT	SUCCESSFUL	,
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Group 1: Hig	h-income OECD Ec	onomies							
(Cyclically Adjusted) Primary Expenditure [30.1]	Mean (Diff w/ t-1) Median Min Max SD #	35.9 36.9 19.0 46.9 6.8	33.8 -2.1 34.1 18.1 45.5 6.8 25 /	32.1 -3.8 32.7 18.0 44.0 6.7 4 / 21	31.6 -4.4 33.1 17.8 42.9 6.5	32.0 32.2 16.4 51.4 8.0	30.6 -1.3 31.1 12.5 49.2 8.3 70 /	31.1 -0.8 31.6 13.3 49.1 8.2 34 / 36	31.6 - 0.4 32.7 11.1 50.3 8.2
(Cyclically Adjusted) Total Revenue [30.5]	Mean (Diff w/ t-1) Median Min Max SD #	33.734.820.143.35.5	34.1 0.5 35.6 18.8 45.0 6.4 25 /	34.6 0.9 36.1 18.8 44.6 7.0 19 / 6	35.3 1.6 37.4 19.5 44.1 7.3	31.3 30.3 16.6 49.4 8.4	32.1 0.7 31.3 16.3 49.1 8.6 70 / 5	32.4 1.1 32.7 16.7 49.1 8.6 2 / 18	32.5 1.2 33.0 16.8 48.8 8.6
Group 3: Mid	dle-income Econo	mies							
(Cyclically Adjusted) Primary Expenditure [25.7]	Mean (Diff w/ t-1) Median Min Max SD #	30.5 29.7 13.4 62.8 11.7	28.0 - 2.6 26.6 10.6 61.8 10.5 54 /	26.5 -4.0 25.1 11.3 52.8 9.0 15 / 39	26.5 -4.0 25.4 11.8 51.3 7.9	27.8 25.9 11.2 55.2 9.9	25.9 -1.8 25.1 10.9 52.0 9.0 159 /	26.7 -1.1 25.8 10.9 52.5 9.0 72 / 87	27.1 -0.7 25.5 12.0 56.5 9.3
(Cyclically Adjusted) Total Revenue [25.1]	Mean (Diff w/ t-1) Median Min Max SD #	26.0 23.9 12.3 51.9 9.2	27.6 1.6 25.7 12.7 52.9 9.5 54 /	27.8 1.8 26.0 15.2 52.9 9.3 39 / 15	28.2 2.2 26.5 15.1 55.2 8.9	26.0 24.1 10.4 54.1 9.8	27.5 1.6 26.2 11.9 54.8 9.8 159 /	26.7 0.7 25.3 11.4 55.2 9.1 (94 / 65	26.7 0.7 25.7 10.8 55.8 9.4

TABLE 3 MOVEMENTS OF PRIMARY EXPENDITURE AND TOTAL REVENUE (AS PERCENT OF GDP)

- (iv) Magnitude of Adjustment—SFA versus NSFA: For Group 1, on the expenditure side, the magnitude of the improvement of primary expenditure across SFA episodes dominates NSFA episodes (-4.4pp versus -0.4pp). The difference is almost negligible for the revenue side (1.6pp versus 1.2pp). On the other hand, for Group 3, the average decreases in primary expenditure are 4.0pp and 0.7pp for SFA and NSFA cases, while the average increases in total revenues are 2.2pp and 0.7pp, respectively. Hence, our results suggest that the revenue side is also important for SFA in EMEs, although the expenditure side still dominates.
- (v) Magnitude of Adjustment in Year t—SFA versus NSFA: Another result, which is closely related to the discussion so far, is that for the two groups, at the initial year t of fiscal adjustment, the decreases in primary expenditure are higher in SFA episodes (for example, 2.1pp for SFA and 1.3pp for NSFA in Group 1). However, the increases in total revenues are higher in NSFA episodes (for example, 0.7% and 0.5% for Group 1) except for Group 3 where the two figures are the same (1.6%). This implies that whether the fiscal adjustment had focused on the expenditure or on the revenue side in the initial year of adjustment, the economies that successfully adjusted tended to focus on decreasing expenditure rather than increasing revenue measures.

In order to assess the importance of sustainability of fiscal adjustment, we repeated the exercise for "successful and sustained" versus "successful and not sustained." Only in Group 3— middle-income economies—was the sample size meaningful and the results very similar to Table 3, SFA vs. NSFA. Moreover, of the 37 episodes of successful adjustment, 24 of them or 62% were sustainable. In both cases, primary expenditure cutbacks dominated increases in total revenue. Not surprisingly, the primary expenditure cutbacks for periods t and t+1 were greater in the case of "successful and not sustained" than in "successful and sustained", suggesting some trade-off between depth and duration of the primary expenditure consolidation process.

C. Components of Expenditure and Revenue

If decreasing expenditure is important for SFA, then is there any particular component in primary expenditure, which plays a more significant role than others? What about such a breakdown by components in total revenue, if any? Table 4 reports the movements of the components of primary expenditure—that is, (i) wages and salaries, (ii) goods and services (excluding wages and salaries), (iii) capital expenditure, and (iv) subsidies.¹² Table 5 reports the components of total revenue—that is, (i) income taxes, (ii) taxes on goods and services, (iii) social security taxes, (iv) taxes on international trade, (v) other taxes, and (vi) nontax revenue.

A smart policymaker may tend to focus on getting the most "bang for the buck" and concentrate efforts on components comprising the highest share of GDP. Hence, (as we will see in Group 1), a priori, we predict that a "larger" component should contribute more to SFA. We also acknowledge that the current literature highlights that EMEs tend to respond to the pressure of fiscal consolidation

¹² Subsidies include all unrequited, nonrepayable transfers on current account to private and public enterprises, and the cost of covering the cash operating deficits of departmental enterprise sales to the public by departmental enterprises.

ITEM		SUCCESSFUL				NOT SUCCESSFUL			
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Group 1: High-income	OECD Economies								
Wages and Salaries	Mean (Diff w/ t-1) #	5.2	4.9 -0.3 24 /	4.6 -0.5 4 / 20	4.6 -0.6	5.2	5.1 -0.1 68 /	5.2 0.0 30 / 38	5.2 0.0
Goods and Services	Mean (Diff w/ t-1) #	4.2	4.0 -0.1 24 /	3.8 -0.3 8 / 16	3.8 -0.4	3.7	3.6 -0.2 68 /	3.7 -0.1 32 / 36	3.7 0.0
Capital Expenditure	Mean (Diff w/ <i>t-1</i>) #	2.5	2.4 -0.2 25 /	2.3 -0.3 6 / 19	2.2 -0.3	3.2	2.9 -0.3 70 /	2.9 -0.3 17 / 53	2.8 -0.3
Subsidies	Mean (Diff w/ <i>t-1</i>) #	23.8	22.3 -1.4 25 /	21.1 -2.7 3 / 22	20.7 -3.0	19.9	19.2 -0.7 70 /	19.5 -0.4 39 / 31	20.0 0.1
Group 3: Middle-incon	ne Economies								
Wages and Salaries	Mean (Diff w/ <i>t-1</i>) #	7.9	7.6 -0.3 54 /	7.4 -0.5 19 / 35	7.3 -0.6	7.6	7.3 - 0.2 158	7.6 -0.0 / 77 / 81	7.7 0.1
Goods and Services	Mean (Diff w/ <i>t-1</i>) #	7.1	6.4 -0.6 54 /	5.9 -1.1 18 / 36	5.6 -1.4	5.7	5.4 -0.3 158	5.3 -0.4 / 67 / 91	5.6 -0.1
Capital Expenditure	Mean (Diff w/ <i>t-1</i>) #	7.4	6.1 -1.3 55 /	5.5 -1.9 15 / 40	5.2 -2.2	6.1	5.3 -0.8 161 ,	5.2 - 0.8 / 62 / 99	5.4 -0.7
Subsidies	Mean (Diff w/ <i>t-1</i>) #	8.1	7.6 -0.5 55 /	7.8 -0.3 24 / 31	7.8 -0.3	8.4	8.1 -0.3 161 ,	8.4 0.0 / 85 / 76	8.4 0.1

 TABLE 4

 MOVEMENTS OF COMPONENTS OF PRIMARY EXPENDITURE (AS PERCENT OF GDP)

Note: Goods and Services do not include Wages and Salaries.

ITEM	ТТЕМ			CESSFUL		NOT SUCCESSFUL			
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Group 1: High-income	OECD Economies								
Income Taxes	Mean (Diff w/ t-1) #	11.0	11.7 0.6 25 /	12.2 1.2 20 / 5	12.2 1.1	9.1	9.5 0.3 70	9.6 0.4 / 44 / 26	9.6 0.5
Taxes on Goods and Services	Mean (Diff w/ t-1) #	11.0	11.1 0.1 25 /	11.3 0.2 18 / 7	11.5 0.5	9.6	9.8 0.2 70	9.9 0.3 / 39 / 31	9.9 0.2
Social Security Taxes	Mean (Diff w/ t-1) #	6.6	6.2 -0.4 25 /	5.9 -0.7 / 13 / 12	6.1 -0.5	7.4	7.5 0.1 70	7.7 0.3 / 49 / 21	7.8 0.4
Taxes on International Trade	Mean (Diff w/ t-1) #	0.3	0.3 0.0 25 /	0.2 0.0 12 / 13	0.2 0.0	0.9	0.9 -0.1 68	0.8 -0.1 / 32 / 36	0.7 -0.2
Other Taxes	Mean (Diff w/ t-1) #	1.2	1.3 0.0 25 /	1.3 0.1 16 / 9	1.5 0.3	1.4	1.5 0.0 70	1.5 0.1 / 41 / 29	1.5 0.1
Nontax Revenue	Mean (Diff w/ t-1) #	3.8	3.9 0.1 25 /	3.8 0.0 / 11 / 14	4.0 0.2	3.0	3.1 0.1 70	3.1 0.1 / 41 / 29	3.1 0.1
Group 3: Middle-incon	ne Economies								
Income Taxes	Mean (Diff w/ t-1) #	5.0	5.5 0.4 54 /	6.0 1.0 / 41 / 13	6.1 1.1	6.1	6.6 0.6 159	6.4 0.3 / 85 / 74	6.3 0.2
Taxes on Goods and Services	Mean (Diff w/ t-1) #	6.3	6.6 0.3 54 /	6.6 0.2 36 / 18	6.8 0.4	5.5	6.0 0.5 159	6.0 0.5 / 107 / 5	6.2 0.6 2
Social Security Taxes	Mean (Diff w/ <i>t-1</i>) #	2.5	2.6 0.0 54 /	2.6 0.1 35 / 19	2.7 0.1	2.8	3.0 0.1 159	3.0 0.1 / 109 / 5	2.9 0.1 0
Taxes on International Trade	Mean (Diff w/ <i>t-1</i>) #	4.5	4.9 0.4 54 /	4.4 -0.1 26 / 28	4.3 -0.2	4.3	4.4 0.0 159	4.2 -0.1 / 66 / 93	4.1 - 0.2
Other Taxes	Mean (Diff w/ <i>t-1</i>) #	1.2	1.2 0.0 54 /	1.2 0.0 31 / 23	1.2 0.1	1.1	1.0 0.0 159	1.0 0.0 / 73 / 86	1.0 - 0.1
Nontax Revenue	Mean (Diff w/ t-1) #	6.6	7.0 0.4 54 /	7.1 0.5 31 / 23	7.3 0.7	6.2	6.6 0.4 159	6.2 0.0 / 83 / 76	6.3 0.1

				TABLE	5				
MOVEMENTS	of Co	MPONENTS	0F	TOTAL	Revenue	(AS	PERCENT	OF	GDP)

ERD WORKING PAPER SERIES No. 79 15

by decreasing *capital expenditure*. This may be politically less costly for various reasons, but on the other hand, public investments need to be safeguarded in EMEs in view of the importance of capital formation on growth and tax and nontax revenues. Can we observe the same phenomenon in our data?

The following results are observed from Tables 4 and 5:

(i) Group 1: In SFA episodes, average primary expenditure declined by about 4.4pp of GDP from t-1 to t+2 (see Table 3). Out of this 4.4pp reduction, the decrease in subsidies account for 3.0pp (see Table 4). Subsidies play an important role possibly (and partly) because they represent a large share of GDP (23.8% across the sample at t-1).¹³

On the revenue side, income taxes—accounting for the largest share of GDP together with taxes on goods and services—contribute relatively more to the revenue increase (by 1.1pp of GDP over 3 years) than other components do (see Table 5). An interesting finding is that the 3-year average increase in social security taxes is negative for SFA episodes (-0.5pp), while it is positive for NSFA ones (0.4pp). While somewhat paradoxical, it may reflect, on the positive side, increased government awareness of the need to avoid taxing social welfare-enhancing activities during periods of fiscal adjustment. Alternatively, it could simply reflect the other side of the reduction in subsidies. That is, in line with cutbacks in subsidies, the government also reduces taxes (although presumably by less than the cutback in spending). Alesina and Perotti (1997) may provide a further clue for this behavior when they refer to the fact that social security contributions like taxes on households tend to have the strongest effects on unit labor costs. A savvy supply-side oriented policymaker would therefore be tempted to target this revenue measure! (However, as we will see later, this may have something to do with the automatic stabilizer properties or sensitivity of social security to GDP growth).

(ii) Group 3: According to Table 3, the reduction in primary expenditure in SFA episodes for Group 3 was 4.0pp. Table 4 suggests that slightly over one half of this improvement is due to the drop in the capital expenditure, which further contributes to the ongoing debate about whether and how to safeguard capital spending in fiscal consolidation strategies. (We revisit this issue in the next subsection.) In fact, as one moves up in income group (from three to one) across successful episodes, reductions in capital spending represent a smaller share of the reduction of the primary deficits. This may suggest that higher- income countries are better able to safeguard capital spending. Turning to goods and services, reductions in spending on this category account for the second largest contribution to successful fiscal adjustment in Group 3.

On the revenue side, one half of the total revenue increase of 2.2pp of GDP (see Table 3) is attributed to the increase in income taxes (1.1pp in Table 5). Furthermore, 0.7pp of the total revenue increase is attributed to nontax revenue. This suggests that policymakers in EMEs are increasing efforts to improve tax and nontax revenues.

¹³ The other three primary expenditure components, wages and salaries, goods and services, and capital expenditure account for 5.2%, 4.2% and 2.5% of the GDP in year *t-1*, respectively, much less than the share of subsidies (23.8%). Group 1's share of subsidies is very large compared to 14.5% and 8.1% (in *t-1*) for Group 2 and Group 3, respectively.

D. Capital Expenditure in EMEs

The results in Table 5 so far point out that over one half of the primary expenditure decrease in SFA episodes of Group 3 is driven, on average, by the cutbacks in capital expenditure.¹⁴ These findings are not inconsistent with other studies on public spending in EMEs, such as, for example, Calderon, Easterly, and Serven (2002b) who argue that in five out of nine Latin American economies in the 1980s and 1990s, infrastructure investment cuts contributed half or more to total fiscal adjustment.¹⁵ To better understand this result we need to drill deeper. In particular, we address whether a heavy reliance on capital expenditure component is in fact commonly observed in EMEs pursuing fiscal adjustment.

First, we examine whether the results are driven by a few influential episodes or whether it tends to be a common occurrence. Table 6 lists all 55 successful episodes of fiscal adjustment in Group 3. According to the table, as many as 31 out of 55 episodes have experienced a decline of capital expenditure of 1pp or more suggesting that decreasing capital expenditure tends to accompany SFA across a majority of EMEs.

Second, we examine the relative importance of the decline in capital spending as a share of the decline in primary spending. Table 6 also shows the decrease (as pp) in primary expenditure from t-1 to t+2 and the *ratio* of capital expenditure cut to primary expenditure cut. From the results, in 19 out of 55 episodes, the decline in capital spending comprised 50% or more of the decline in primary expenditure. Therefore, we interpret that our analysis in the previous subsection on the capital expenditure is *not* driven by a few influential episodes.

E. Expenditure on Interests

Primary expenditure, by definition, excludes government expenditure on interests. However, a SFA has an indirect impact on interest expenditure, mainly because it tends initially to decrease the growth of government debt (and perhaps even decrease the stock of debt) and hence the government can reduce the deadweight loss and free up more resources on noninterest payments. Therefore, we predict that during SFA episodes the growth of expenditure on interests should at least slow down (especially if debt level is high), if not, decrease.

¹⁴ As "capital expenditure" is largely contributing to "public investment" particularly in EMEs, some economists have argued that it should be given special consideration in the analysis of fiscal space creation and should not be treated simply as "one more component" of primary expenditure (for an analysis of whether fiscal deficit targets should or should not include public investment, see IMF 2004). As is observed in the result, it may be questionable to assert that a decline in primary expenditure resulting from a reduction in capital expenditure is considered a "successful" fiscal adjustment. Therefore, in a later subsection, we try to introduce a new definition on (successful) fiscal adjustments for developing economies.

¹⁵ While it could be argued that with liberalization, private investment in infrastructure could make up for the shortfall in public infrastructure investment. This does not seem to be the case. According to Calderon, Easterly, and Serven (2002a, 20), "...there is only limited evidence to support the common perception that privatization ... explains the observed downward trend in public infrastructure spending."

EPISODE	CAP] t-1	TAL EXPE	NDITURE t+1	(% OF 0 <i>t+2</i>	GDP) CHANGE (A)*	CHANGE IN PRI. EXP. (B)*	*(A) / (B) X 100
Oman 1976	21.4	20.6	14.1	9.3	-12.0	-11.5	104.2
Oman 1977	20.6	14.1	9.3	9.8	-10.7	-21.1	50.8
Malaysia 1982	15.1	10.9	8.1	5.4	-9.7	-13.1	74.2
St. Kitts and Nevis 1989	12.6	8.6	3.4	4.0	-8.6	-8.7	98.8
Jamaica 1997	13.7	6.9	5.2	5.1	-8.5	-6.7	126.7
Malaysia 1983	10.9	8.1	5.4	2.7	-8.2	-11.2	72.8
Egypt, Arab Rep. 1977	16.4	8.4	6.9	8.4	-7.9	-13.7	57.8
Oman 1987	13.4	8.0	7.1	5.8	-7.6	-13.4	56.6
Swaziland 1987	7.7	4.3	8.0	0.9	-6.8	-24.6	27.6
Morocco 1982	12.9	12.7	8.0	6.4	-6.4	-9.8	65.5
Jordan 1983	12.6	11.0	8.7	7.4	-5.2	-1.3	411.2
Namibia 1988	9.4	7.0	4.1	5.0	-4.4	-7.3	59.7
Iran, Islamic Rep. 1983	8.7	8.3	5.6	4.7	-4.0	-9.2	43.2
Sri Lanka 1981	16.6	13.2	15.5	13.0	-3.6	-12.3	29.3
Sri Lanka 1989	10.3	8.2	6.1	6.8	-3.5	-2.1	169.3
Botswana 1983	10.2	6.7	6.4	6.7	-3.5	-7.5	46.4
Oman 1995	6.9	6.4	3.9	3.7	-3.2	-13.5	23.8
Hungary 1994	7.6	6.1	4.7	4.6	-3.0	-14.7	20.6
South Africa 1994	3.5	2.0	1.7	1.4	-2.1	-3.1	67.7
Botswana 1977	11.2	10.0	12.0	9.2	-1.9	-4.9	39.9
Belarus 1998	6.6	4.2	6.4	4.9	-1.7	-6.8	24.9
Jordan 1992	7.5	6.9	5.0	6.0	-1.6	-3.6	43.6
Bolivia 1995	5.0	4.1	5.0	3.6	-1.4	-0.5	267.8
Chile 1977	4.9	3.9	3.9	3.5	-1.4	-2.8	50.2
Jamaica 1998	6.9	5.2	5.1	5.5	-1.4	-2.3	59.6
Costa Rica 1981	5.2	3.5	2.5	4.0	-1.3	-2.6	47.7
Chile 1978	3.9	3.9	3.5	2.7	-1.2	-3.0	39.9
Tunisia 1979	11.7	9.1	9.5	10.5	-1.2	-1.2	97.0
Hungary 1995	6.1	4./	4.6	5.0	-1.1	-13.0	8.7
Peru 1995	4.3	2.8	2.2	3.2	-1.1	-0.7	149.0
Morocco 1990	7.9	8.0	6.1	6.9	-1.0	0.1	N/A#
Inailand 1988	3.2	2.0	2.0	2.5	-0.8	-4.7	16.2
Peru 1996	2.8	2.2	3.2	2.2	-0.6	-1.2	52.8
Mexico 1995	2.0	1.5	2.3	2.0	-0.6	0.0	N/A
Uruguay 1990	2.0	1./	1.4	1.4	-0.0	2.0	N/A
Politica 100/	Z.Z	1.5	1.2	1.0	-0.6	-1.0	24.9
Bolivia 1994	5.4	5.0	4.1	5.0	-0.4	-1.4	52.0
Jordan 1000	5.9	6.9	4.5	5.5	-0.4	-2.1	N/A
Mauritius 1983	/.0	2.2	2.5	2.0	-0.2	-2.6	3.3
Chile 1990	2.8	2.0	2.5	3.9	0.2	-2.0	N/A
Thailand 1980	2.6	2.1	2.5	2.8	0.2	-2.5	-9.0
Mauritius 1986	2.0	2.0	1.1.	/. 2	0.2	_1.3	-18 /
Chile 1993	3.9	3.0	4.4	4.2	0.2	0.1	N/A
Botswana 1984	6.7	6.4	6.7	7 1	0.3	3.2	N/A
	0.7	0.7	5.7	/.1	0.5	5.2	,,,,

 Table 6

 Movements of Capital Expenditures (as percent of GDP) in Individual Successful Episodes in Group 3

continued.

Turkey 1997	2.1	3.4	2.3	2.7	0.5	0.1	N/A
Belarus 1997	5.5	6.6	4.2	6.4	0.8	-8.7	-9.5
Venezuela, RB 1973	4.4	3.6	6.4	5.7	1.3	1.8	N/A
Uruguay 1976	1.3	2.3	2.6	2.6	1.4	-0.5	-256.7
Morocco 1988	6.2	7.0	7.9	8.0	1.9	0.3	N/A
Iran, Islamic Rep. 1990	3.1	5.0	4.9	5.0	1.9	1.9	N/A
Croatia 1995	2.7	3.6	5.4	4.6	1.9	5.6	N/A
Egypt, Arab Rep. 1991	4.8	5.6	12.1	6.7	2.0	3.5	N/A
Swaziland 1975	8.2	9.7	5.0	10.8	2.5	7.3	N/A
Iran, Islamic Rep. 1991	5.0	4.9	5.0	7.6	2.7	3.9	N/A

* "Change" implies the difference between the observed values of t-1 and t+2.

** "Change" on Primary Expenditure, as percent of GDP.

"N/A" in this column implies that primary expenditure has not decreased from t-1 to t+2.

Table 7 reports the movement of expenditure on interests during fiscal adjustment.

- (i) *Group 1*: For Group 1, expenditure on interests has increased from 4.7% of GDP to 5.1% from *t*-1 to *t*, and remains constant until *t*+2. "Not successful" cases reveal a very small increase during the same period.
- (ii) Group 3: A similar trend is observed in Group 3. From t to t+2, expenditure on interests in successful episodes (where government debt was significantly higher than in "not successful" episodes) has moderately increased by 0.1pp of GDP, while it has increased by a larger 0.4pp in "not successful" episodes.

ITEM		SUC	CESSFUL			NOT	SUCCESSF	UL	
	t-1	t	t+1	t+2	t-1	t	t+1	t+2	
Group 1: High-income OECD Ec	onomies	[3.1]							
Mean (Diff w/ t) Median Min Max SD	4.7 4.4 0.4 10.3 2.3	5.1 4.6 0.4 10.3 2.6	5.1 0.0 5.1 0.5 9.3 2.7	5.1 0.0 3.9 0.6 10.9 2.9	3.1 1.9 0.0 11.3 3.0	3.4 2.4 0.1 13.7 3.1	3.4 0.0 2.3 0.2 12.3 3.0	3.5 0.1 2.4 0.2 13.7 3.1	
#		25 /	13 / 12			70 /	45 / 25		
Group 3: Middle-income Econor	nies [2.	9]							
Mean (Diff w/ t) Median Min Max SD #	2.9 2.3 0.0 12.5 2.4	3.1 2.3 0.0 13.3 2.8 55 /	3.3 0.2 2.3 0.0 14.9 3.2 28 / 27	3.2 0.1 1.7 0.0 14.9 3.5	2.6 1.9 0.0 11.0 2.1	2.8 2.1 0.0 15.1 2.7 161	2.9 0.1 2.3 0.0 19.8 2.9 / 101 / 60	3.3 0.4 2.5 0.0 29.2 3.7	

 TABLE 7

 MOVEMENTS OF EXPENDITURE ON INTEREST (AS PERCENT OF GDP)

(iii) *Initial Conditions*: For all two groups, the initial level of expenditure on interests is higher in SFA episodes than in NSFA episodes. This suggests that initial conditions—reflecting higher interest expenditure—are likely to positively affect the probability of a SFA.

F. Removing Capital Expenditure

As discussed above, approximately one half of the decrease in primary expenditure in SFA in EMEs is on average due to a decrease in capital expenditure. However, capital expenditure is critical especially in EMEs, given the importance of capital formation for economic growth, especially in the presence of limited private sector investment. In this context, our current definitions of SFA does not distinguish between a "smart" fiscal adjustment—whereby the government "saves" by decreasing noninvestment expenditure and/or increasing total revenue *and* reallocates these savings for public investment—and a conventional adjustment.

To address this issue, we introduce an alternative definition, namely the "primary deficit net of capital expenditure" to replace the "primary deficit." All the other definitions are the same as before. Table 8 is the duplication of previous Tables 1-5 and 7 using the alternative definition for Group 3 where reductions in capital spending were contributing to fiscal adjustment. Although the numbers are different, most of the conclusions we have derived so far still hold for this new definition. Hence, whether we include capital expenditure in defining fiscal adjustment or not does not seem to affect our main arguments.

V. MACROECONOMIC ADJUSTMENT

In the previous section, we described salient features of SFA across category of countries by examining the behavior of *fiscal* variables. In this section, we focus on the relation between fiscal adjustment and economic growth and, more generally, *macroeconomic* variables. In particular, we examine what explains either expansionary or contractionary fiscal adjustment. The first subsection reviews possible explanations on why fiscal adjustment can be expansionary. The second and third subsections present empirical results on the evolution of GDP growth, unemployment, and the national accounts components of GDP over periods of fiscal adjustment. The fourth subsection discusses whether initial conditions on fiscal variables affect the movement of macroeconomic variables; that is, we try to answer, "If initial fiscal conditions are worse, is SFA accompanied by different macroeconomic performance?" The final subsection discusses the probit result in order to answer which explanatory variables account for the success probability of fiscal adjustment.

A. Why Fiscal Adjustment is Not Necessarily Contractionary

In the standard Keynesian IS-LM framework, fiscal consolidation—whether through a reduction in public spending or an increase in the tax rate—is contractionary. Aggregate demand declines due to (i) lower public spending (and consequent impact through the Keynesian multiplier) or (ii) lower private spending as higher taxes reduce household disposable income. In the goods market, as demand for goods declines, equilibrium output declines. In the money markets, income declines and therefore the quantity of money demanded is lower. There is excess supply for real balances and in equilibrium interest rates decline. Consequently, following fiscal consolidation, the goods and money markets clear at a lower (level of) equilibrium income and interest rates with planned spending equal to income, at the same time, the quantity of real money balances demanded equal to the given real money stock.

However, this standard IS-LM construct faces various criticisms. The first critique is that the framework is essentially static whereas macroeconomic systems are intrinsically dynamic. This has repercussions especially as these short-run macro models do not incorporate the role of capital accumulation in the macro-dynamic analysis. Instead the focus is on the demand side of investment. As a consequence, the supply-side response of firms and its impact on the stock of capital in the economy and the resulting productive capacity is largely unaccounted for.

The second critique is that agents' expectations in these early models take a back-seat role and when they figure more prominently these tend to be associated with adaptive expectations hypothesis. In other words, expectations are formed by looking at past evolution of the relevant variable. Embedding alternative rational expectations hypothesis, that is, assuming that on average forecasters are correct has significant implications on macroeconomic theory particularly on the analysis of macroeconomic policy (see Turnovsky 2000).

Returning to the impact of fiscal consolidation, empirical studies (for example, Alesina and Ardagna 1998; and Alesina, Perotti, and Tavares 1998) have reported that many episodes of fiscal adjustment were indeed expansionary. These non-Keynesian results have been reconciled according to one of the following hypotheses.

1. Consumption-Expectations Hypothesis

A first hypothesis is derived from a longer-term analysis based on forward-looking expectations and the degree of credibility of the policymaker. In particular, following an announcement of fiscal consolidation—and assuming forward-looking households indeed believe the announcement to be credible—the resulting reduction in public spending may lead to an expectation of lower taxes and higher disposable income over the medium term and lead to changes in consumption behavior today.¹⁶ In the extreme case, if agents were to believe that fiscal adjustment was to render the decrease in government spending permanently, then there could be a more robust change in behavior.¹⁷ In this case, growth in consumption could jump in order to mimic the perceived sudden growth in life-time after tax "permanent" income. It may be argued that following a country's entry

¹⁶ This is predicated on the underlying assumptions of (i) the time horizon such that households have finite horizons (and bequests are not fully transferable) in contrast to the dynastic households with fully transferable bequests where Ricardian equivalence would hold; (ii) public and private consumption are not perfect substitutes; and/or (iii) taxes are distortionary.

¹⁷ There may be a further argument to explain the non-Keynesian results. Indeed if taxes are distortionary, a decrease in public spending may usher in a period of lower taxes, thereby reducing tax distortions in the economy, *ceteris paribus*, leading to an expansion of economic activity through a more efficient allocation of resources.

ITEM		SUC	CESSFUL			NOT	SUCCESS	SFUL	
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Table 1(CyclicallyAdjusted)PrimaryDeficit minusCapitalExpenditure[-5.0]	Mean (Diff w/ t-1) Median Min Max SD #	-1.9 -2.2 -15.9 15.0 6.6	-6.9 -5.0 -6.0 -21.3 3.7 5.3 49	-7.6 -5.7 -7.2 -28.4 3.1 6.3 / 6 / 43	-8.0 -6.1 -7.1 -28.8 11.7 7.0	-3.3 -2.6 -24.1 11.6 5.6	-7.7 -4.4 -6.8 -28.4 2.1 5.6 155	-5.1 -1.8 -4.5 -28.8 15.0 5.9 / 51 / 1	-4.8 -1.4 -4.4 -24.6 10.4 5.3 04
Table 2 Government Debt [43.6]	Mean (Diff w/ t-1) Median Min Max SD #	52.4 40.5 3.0 134.7 38.1	55.4 3.0 44.3 2.6 152.4 38.8 35	49.4 - 3.0 38.3 1.6 152.4 38.1 / 12 / 23	46.2 -6.3 31.7 2.6 152.4 37.2	38.8 31.3 0.0 243.8 34.4	40.0 1.3 32.3 0.0 243.8 39.6 79	42.8 4.0 35.0 0.0 332.1 46.5 / 57 / 22	49.0 10.3 36.7 0.0 401.0 58.5
Table 3(CyclicallyAdjusted)PrimaryExpenditureminus CapitalExpenditure[20.1](CyclicallyAdjusted)TotalRevenue[25.1]	Mean (Diff w/ t-1) Median Min Max SD # Mean (Diff w/ t-1) Median Min Max SD #	23.2 21.1 8.8 45.6 9.3 26.4 23.2 11.9 51.9 9.7	21.7 -1.5 19.9 6.0 44.9 9.1 49 28.3 1.9 26.1 11.7 52.9 9.9 49 49	21.4 -1.8 20.6 8.1 43.1 8.5 / 14 / 35 28.8 2.4 26.3 15.2 52.9 9.9 / 36 / 13	21.2 -2.0 19.8 6.2 41.9 8.0 29.5 3.1 26.7 13.6 55.2 9.7	21.4 20.3 5.1 51.3 8.7 25.5 24.1 10.4 54.1 9.6	20.5 -0.9 19.6 8.2 48.3 7.9 155 27.5 2.1 26.3 11.9 54.8 9.7 155	21.1 -0.3 20.0 7.4 48.7 7.9 / 76 / 7 26.3 0.8 25.3 10.7 55.2 9.1 / 91 / 6	21.4 0.0 20.4 7.9 52.4 8.2 9 26.0 0.5 25.7 10.4 55.8 9.2 4
Table 4: Compo Wages and Salaries	nents of Primary Mean (Diff w/ t-1) #	Expendit 7.6	ture 7.3 -0.3 50 /	7.3 -0.3 / 18 / 32	7.1 -0.5	7.5	7.4 -0.1 152	7.7 0.1 / 78 / 7	7.8 0.2 4
Goods and Services*	Mean (Diff w/ <i>t-1</i>) #	7.0	6.3 -0.7 50 /	6.1 -0.9 / 17 / 33	5.7 -1.2	5.7	5.3 -0.3 152	5.3 -0.4 / 70 / 8	5.5 -0.2 2
Capital Expenditure	Mean (Diff w/ <i>t-1</i>) #	6.6	6.1 -0.5 50 /	5.9 -0.7 / 23 / 27	5.9 -0.6	6.0	6.1 0.1 158	5.8 -0.2 / 72 / 8	5.5 -0.5 6

 TABLE 8

 Duplicated Tables Excluding Capital Expenditure for Group 3 (as percent of GDP)

ITEM			SUC	CESSFUL			NOT	SUCCESS	FUL
		t-1	t	t+1	t+2	t-1	t	t+1	t+2
Subsidies	Mean (Diff w/ <i>t-1</i>) #	8.7	7.9 -0.8 50 /	8.2 -0.5 20 / 30	8.0 - 0.7	8.1	7.8 -0.3 158	8.1 0.0 / 80 / 78	8.1 0.0
Table 5: Comp	onents of Total Rev	enue							
Income Taxes	Mean (Diff w/ <i>t-1</i>) #	4.7	5.3 0.6 49 /	5.8 1.1 37 / 12	6.0 1.3	6.0	6.6 0.6 155	6.4 0.3 / 92 / 63	6.3 0.2
Taxes on Goods and Services	Mean (Diff w/ <i>t-1</i>) #	6.0	6.3 0.3 49 /	6.3 0.2 32 / 17	6.5 0.4	5.5	6.1 0.6 154	5.9 0.4 / 103 / 5	6.1 0.6 1
Social Security Taxes	Mean (Diff w/ t-1) #	3.0	3.0 0.0 49 /	3.0 0.0 32 / 17	3.1 0.0	2.7	2.8 0.1 155	2.8 0.1 / 107 / 4	2.8 0.1
Taxes on International Trade	Mean (Diff w/ t-1) #	4.8	5.2 0.4 49 /	4.7 -0.1 29 / 20	4.7 -0.1	4.5	4.7 0.3 154	4.4 -0.1 / 67 / 87	4.3 -0.2
Other Taxes	Mean (Diff w/ t-1) #	1.3	1.3 0.0 49 /	1.4 0.1 30 / 19	1.4 0.1	1.0	1.0 0.0 155	1.0 -0.1 / 71 / 84	0.9 -0.1
Nontax Revenue	Mean (Diff w/ <i>t-1</i>) #	6.6	7.2 0.6 49 /	7.7 1.1 33 / 16	8.0 1.4	5.8	6.3 0.5 155	5.9 0.1 / 80 / 75	5.8 -0.1
Table 6: Expen	diture on Interests								
Expenditure on Interests	Mean (Diff w/ t) Median Min Max SD #	2.7 2.2 0.0 9.4 2.2	2.8 2.0 0.0 10.3 2.5 50 /	2.9 0.1 2.1 0.0 11.6 2.8 25 / 25	2.9 0.2 1.6 0.0 13.5 3.2	2.5 1.8 0.0 14.6 2.4	2.6 1.7 0.0 15.1 2.8 158	2.9 0.2 1.9 0.0 19.8 3.3 / 98 / 60	3.1 0.5 2.1 0.0 29.2 3.8
	π		50 /	25 / 25			150	, 50 , 00	

Note: Goods and Services do not include Wages and Salaries.

into OECD, EU, Euro Area (EA), or other economic clubs requiring fiscal discipline, there might be some element of greater credibility, perhaps giving an additional boost to private consumption.

2. Supply-side-plus Hypothesis

Another hypothesis considers in more detail the impact of supply-side effects from fiscal consolidation on investment and capital formation. In the case of a credible government commitment to fiscal consolidation through lower public spending, forward-looking firms could react by increasing investment today on account of (i) lower perceived future taxes (a similar argument to the above on the investment side) and (ii) longer-term implications for firms and households' cost of funds as government demand on loanable funds declines.¹⁸ This could be considered a first phase of adjustment.

However, there could be a more significant second round effect stemming from the reaction of financial markets in an open economy context to the perceived sustainability of the fiscal consolidation measures. Indeed if improved fiscal conditions were to lower the country risk premium as usually happens, there could be significant capital flows into the economy leading to increased investment and eventually higher potential GDP growth. This would particularly hold true for EMEs where capital flows may be volatile and where access to external funding and overall high risk premium may represent binding constraints to improved investment and economic growth.

3. More Supply Side: Unit Cost of Labor

Fiscal consolidation may further stimulate the economy by decreasing unit cost of labor. Compared to "not successful" episodes, successful ones are on average accompanied by a larger expenditure cutback—or containment in the growth—of "wages and salaries" component (see Table 4.) This decline in growth of public sector wages may spillover to the rest of the economy resulting in a (relative) decline in the unit cost of labor and labor market equilibrium wages, which in turn reduces firms' costs of production. Empirically, Alesina and Ardagna (1998) find that wages and salaries as a share of GDP has decreased on average after successful fiscal adjustment in OECD economies.¹⁹

B. GDP Growth Rate and Unemployment Rate

Against the theoretical underpinnings in the previous section, we now turn to the empirical evidence to answer whether SFA is indeed expansionary. To answer this question, we first need to identify a proxy for business cycle. In this paper, we use the GDP growth rate net of its "trend",

¹⁸ In this instance, changes in the yield curve could capture the perceived market sentiment.

¹⁹ However, we find it is not possible to confirm this argument with EMEs because of data limitations on unit labor cost.

which is obtained by the Hodrick-Prescott filter.²⁰ As is shown in Figure 1, the filtered series shows the "trend" growth rate, reflecting an economy's possible "*mid-term* change" over time. Hence, our indicator (or "normalized" growth) reflects short-term changes, and also has a sample mean equal to zero for each country.



FIGURE 1 EXAMPLE OF HODRICK-PRESCOTT FILTER (JAPAN)

This figure shows the movements of the actual GDP and its Hodrick-Prescott filtered series from 1970 to 2002 in Japan. The difference between these two is used as a proxy for a business cycle.

The descriptive statistics of our business cycle proxy ("normalized" growth) are summarized in Table 9. Figure 2, based on the figures reported in Table 9, provides a clearer depiction of the evolution of the proxy. The left figures in Figure 2 show the *means* normalized GDP growth rates while the right ones show the *first, second and third quartiles of the same variable*.

The following results are observed from Table 9 and Figure 2:

(i) Movement of Mean in Year t-2, t-1: One year prior to fiscal adjustment (in year t-1), the sample mean for our proxy stays at a peak of about 0.9% for high-income group and 1.5% for middle-income group, compared to a mean close to yet below zero in year t-2 for both groups. This implies that the economy is, on average, expanding right before the fiscal adjustment. Furthermore, in SFA the sample mean in year t-1 is never below that

²⁰ This proxy was already introduced previously when we defined the cyclical adjustment.

	\sim
	TREND
	ITS
	OF
	NET
	RATE
	GROWTH
	GDP
	(real
BLE 9	RATE
TAI	GROWTH
	GDP
	"NORMALIZED"
	THE
	. 10
	MOVEMENT

ITEM				SUC	CESSFUL						NOT	SUCCESS	FUL			
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1: Hiç	gh-income	e OECD E	conomies													
Mean	-0.2	0.9	1.0	0.6	-0.3	0.2	0.1	0.3	-0.5	0.9	-0.1	-0.1	0.3	-0.3	-0.2	-0.1
25%	-1.2	0.0	0.1	-0.2	-1.5	-0.6	-1.0	-0.6	-1.8	-0.2	-1.4	-1.5	-0.8	-1.3	-2.0	-1.0
Median	0.0	1.1	1.3	0.8	0.0	0.1	0.0	0.1	-0.3	1.1	0.1	0.3	0.2	-0.2	0.2	0.0
75%	1.1	2.1	2.0	1.8	0.8	1.4	1.0	1.6	0.8	2.5	1.4	1.5	1.6	1.0	1.2	1.1
Min	-3.6	-3.3	-2.5	-3.0	-8.8	-3.7	-2.1	-2.6	-8.8	-5.5	-5.5	-8.8	-5.5	-9.4	-5.0	-5.5
Мах	4.0	3.5	5.4	2.4	3.4	4.8	3.0	2.7	3.8	0.0	3.9	4.8	3.9	3.7	3.9	3.9
SD	2.2	1.7	1.8	1.4	2.4	2.0	1.5	1.5	2.3	2.3	2.0	2.4	2.0	2.3	2.0	1.9
#					25							71				
Group 3: Mie	ddle-incor	me Econo	mies													
Mean	-0.3	1.5	1.3	0.5	0.2	0.2	-0.2	-0.3	-0.2	0.9	-0.3	0.4	-0.4	-0.5	0.0	-0.3
25%	-2.6	-0.9	-2.0	-1.5	-1.8	-1.8	-2.1	-2.7	-2.5	-1.4	-2.2	-1.5	-2.3	-2.4	-2.1	-2.1
Median	0.3	0.8	0.7	0.2	0.3	0.4	0.4	0.6	-0.3	1.1	-0.2	0.6	0.1	0.2	0.2	-0.1
75%	1.6	3.4	4.7	2.3	2.4	2.3	2.0	1.8	2.2	3.4	2.1	2.4	2.1	1.8	2.4	1.7
Min	-13.0	-15.5	-10.8	-12.5	-12.5	-10.5	-13.8	-13.8	-15.4	-12.5	-13.0	-11.4	-13.8	-15.5	-12.8	-15.5
Мах	16.3	16.3	15.7	31.5	15.7	10.1	9.9	9.9	14.5	14.7	13.5	9.5	8.5	9.9	7.8	14.5
SD	4.6	5.2	5.5	5.8	4.2	3.5	4.1	4.2	4.5	3.9	3.9	3.8	4.1	4.1	3.7	4.1
#					53							153				

Note: See the text for definition of normalized GDP growth rate.



Group 1: Mean (Left) and three Quartiles (Right)



*Two bold lines denote the median.





*Two bold lines denote the median.

of NSFA. Therefore, we may "expect" that the better the cycle is in t-1, the higher the probability of SFA in t or that the timing of the introduction of a fiscal adjustment matters.

- (ii) Movement of Mean in Year t: In the year of fiscal adjustment (in year t), our proxy in SFA episodes has a mean greater than or equal to 1% for both Groups 1 and 3, while it drops to about the proxy's average level, 0%, in NSFA episodes in Groups 1 and 3. A possible interpretation is that if economic growth remains buoyant in year t, then the greater the likelihood that the government will continue its fiscal reform in the following years.
- (iii) Movement of Mean in Year t+1 and Later: In t+1, the mean stays relatively higher in SFA episodes than in NSFA ones, at least for Group 1. Furthermore, in Group 1, SFA episodes have higher mean values than NSFA in all periods up to t+5 except for t+2. The mean value in those episodes for Group 3 is higher than or similar to the one in NSFA episodes. (The median in SFA cases in Group 3 is steadily increasing from t+1 to t+5, and always higher than the one in NSFA cases from t+2 onward.) In NSFA episodes, the mean (which has already dropped to about 0% in year t) seems to stay around or below 0% although there are some fluctuations. For Group 1, there is an increase of the mean from -0.3% to 0.2% in year t+3, but we do not think it contains significant information if the observed mean in t+2 (-0.3%) is too low, perhaps because of small sample size. For Group 3, the mean level is declining from t+1 to t+5: this may be simply a coincidence or it may reflect that the economy is deteriorating because deficits are accumulating (we will come back to this issue later).

Presenting only the mean might be misleading especially when outliers exist. This may be true in our case—we observe a maximum of 31.5% (in t+1 for SFA Group 3), which may affect the sample mean value and a minimum of -15.5% in NSFA for Group 3. However, as seen in the right side diagram in Figure 2, where quartiles are presented (for 25%, 50%, 75% range) these outliers do not change the time trend significantly.

According to the above results, it is not clear whether fiscal adjustment tends to support "sustainable" growth of the economy over the medium term. Alternatively, we consider whether the unemployment rate may be affected by fiscal adjustment. As noted, a problem using this variable is that for EMEs there are cases of missing data. Accordingly, we focus on the evolution of unemployment rates across MEs. Table 10 and Figure 3 report its movement for Group 1 and Group 3. We do caveat that due to the few observations remaining once missing data cases are removed (i.e., only 13 SFA episodes, in contrast to 53 in Table 9), our results for Groups 3 cannot be considered robust.

(iv) Unemployment Rate:²¹ For Group 1, the mean and the median observations point to a SFA that appears to lead to a continuous decrease of the unemployment rate (from 8.4% in t to 6.7% in t+5). On the other hand, in NSFA episodes, the unemployment rates in t and t+5 are rather similar (despite a small decline in between. Therefore, SFA does appear to have an impact on the economy—it decreases the unemployment

²¹ The unemployment rate is not normalized because we do not have sufficient observations to apply appropriate regressions for Hodrick-Prescott filtering, especially for EMEs.

ITEM			SU	CCESS	FUL						NOT	SUCC	ESSFU	L		
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1: Hig	gh-inco	me OE	CD Eco	onomie	es											
Mean	8.8	8.8	8.4	7.8	7.5	7.2	6.9	6.7	7.2	7.3	7.2	7.0	6.9	6.9	7.1	7.3
25%	6.8	7.3	6.7	6.1	6.1	6.1	6.1	5.7	3.9	3.5	3.5	2.9	3.2	3.1	3.4	3.5
Median	9.5	9.5	8.9	8.0	7.7	7.3	7.0	7.2	7.1	7.0	6.9	6.1	5.8	6.8	7.0	6.5
75%	10.4	10.2	9.6	9.1	9.5	9.0	9.0	8.7	8.9	9.0	9.9	9.2	9.4	9.2	9.7	10.5
Min	1.0	1.3	3.2	2.7	2.9	2.3	1.9	1.6	0.7	0.4	0.6	1.6	1.5	1.6	1.5	1.6
Max	16.4	16.6	15.5	14.6	12.7	11.5	11.6	10.3	19.6	21.0	20.6	19.9	18.9	17.2	17.0	16.4
SD	3.4	3.3	3.0	3.0	2.8	2.8	2.8	2.7	4.4	4.6	4.4	4.4	4.4	4.3	4.3	4.4
#					20								41			
Group 3: Mi	ddle-ind	ome E	conon	nies												
Mean	7.4	7.1	6.7	6.4	6.3	6.0	6.1	6.5	8.8	9.1	9.1	8.6	8.3	8.6	8.4	8.7
25%	5.3	4.4	4.5	4.3	4.0	2.9	2.3	2.2	5.9	5.6	5.5	5.9	5.7	5.9	5.9	5.6
Median	7.5	6.5	6.7	6.8	7.7	6.6	7.0	7.0	8.1	8.6	8.7	8.7	8.3	8.7	7.7	7.9
75%	9.9	8.9	8.5	9.0	8.7	8.0	8.0	7.9	11.1	13.0	11.8	10.4	11.0	11.4	11.2	11.4
Min	0.5	1.4	1.4	1.4	2.2	1.4	1.4	1.3	1.3	2.8	2.9	2.4	3.4	2.4	3.0	3.1
Max	14.8	14.5	14.5	10.2	9.9	11.4	13.5	16.1	19.9	21.4	20.0	16.3	15.6	16.0	16.1	18.2
SD	3.9	3.7	3.7	3.0	2.8	3.0	3.6	4.3	4.2	4.2	3.9	3.6	3.3	3.4	3.4	4.0
#					13								35			

TABLE 10MOVEMENT OF THE UNEMPLOYMENT RATE

rate steadily. Results from Group 3—subject to the small observations caveat—also support this finding. The mean value in SFA episodes decreases marginally from 6.7% in t to 6.0% in t+3 (and then increases to 6.5% in t+5). However, we do note that for Group 3 in NSFA episodes the mean unemployment rate also declines from 9.1% in t to 8.7% in t+5.

1. Is Monetary Policy Influencing the Results?

Bearing in mind the possible effect of monetary policy on equilibrium output over the short term, we examine below this consideration. In particular, to what extent can easy money resulting from a more accommodating monetary policy stance drive the adjustment process in the economy and lead to a temporary expansion in the economy over and beyond the impact from fiscal consolidation? If fiscal adjustment is in fact accompanied by expansionary monetary policy, one needs to be more careful about the interpretation of the results. Table 11 and Figure 4 report the movements of inflation rates (measured by the GDP deflator) during periods of fiscal adjustment. We observe that (i) the inflation rate is lower in successful episodes and (ii) in both SFA and NSFA episodes for both Groups 1 and 3, the inflation rates are steadily declining over time.²²

²² Notice that the mean of inflation rate for Group 3 is not representative because it is driven by some hyperinflation cases. (The median can be more representative in this case.)







*Two bold lines denote the median.





*Two bold lines denote the median.

TABLE 11 MOVEMENT OF THE INFLATION RATE (MEASURED BY GDP DEFLATOR, %)

ITEM				SUCC	ESSFUL							NOT	SUCCESS	FUL		
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1: Hi	gh-incom	e OECD	Economies	: [7.3]												
Mean	5.8	6.3	7.0	6.8	5.5	4.1	3.9	4.0	10.0	10.1	10.1	8.9	8.6	8.1	8.1	7.7
25%	2.3	2.3	2.8	2.6	2.1	1.7	1.2	2.4	3.7	3.5	3.9	3.0	2.9	2.8	2.7	2.4
Median	5.7	5.7	6.0	4.6	3.3	4.0	4.0	2.9	6.8	7.5	7.0	6.1	5.4	5.6	5.2	4.5
75%	8.0	8.4	8.4	10.1	7.5	7.0	5.2	5.1	14.8	14.4	12.6	11.0	10.5	8.8	10.1	8.8
Min	-0.8	-0.1	1.2	-0.5	-0.9	-0.7	-0.3	-1.2	-0.9	-0.1	0.1	-1.2	-0.7	0.7	0.5	-0.3
Мах	16.7	20.7	20.7	19.8	18.4	14.4	11.2	15.9	40.9	52.5	49.8	53.3	76.2	49.8	53.3	76.2
SD	4.4	5.0	5.5	6.0	5.0	4.1	3.0	3.5	8.7	9.2	9.7	9.3	11.5	8.8	0.0	11.2
#				25								71				
Group 3: Mi	iddle-inco	me Ecol	nomies [5	4.9] [meo	lian: 8.3]											
Mean	89.5	43.2	57.3	33.2	22.3	19.6	15.6	18.9	24.5	29.2	37.4	21.1	20.9	29.7	30.0	39.5
25%	6.9	9.0	9.0	6.5	5.8	6.2	3.8	4.0	6.0	5.3	7.1	4.7	5.0	4.4	4.5	4.5
Median	13.1	11.8	11.2	12.1	10.6	10.6	9.7	8.0	12.1	12.1	15.5	10.6	9.8	9.7	10.4	9.0
75%	22.4	19.5	21.2	23.6	22.3	18.0	13.8	17.0	22.3	23.6	25.7	22.1	18.5	22.7	20.5	21.0
Min	-7.7	-20.6	1.3	-8.1	-2.4	-13.4	-11.4	-6.7	-8.6	-8.1	-2.4	-13.4	-13.4	-7.7	-10.0	-20.6
Max	1466.8	1053.4	1945.1	661.5	316.8	185.3	79.6	316.8	414.0	665.4	1466.8	250.6	231.7	948.5	1322.5	2509.5
SD	261.7	147.2	265.5	91.2	44.5	30.2	20.7	45.1	43.2	75.7	125.0	34.4	36.0	94.7	111.1	209.4
#				53								153				





Group 1: Mean (Left) and three Quartiles (Right)

*Two bold lines denote the median.



Group 3: Mean (Left) and three Quartiles (Right)

*Two bold lines denote the median.

Table 12 and Figure 5 also show the movements of money and quasi-money growth rate, which is a better indicator of the monetary policy stance. Observing the movement of medians, successful episodes in Group 1 do not reveal any severe tendency of expansionary monetary policy.²³ Similarly in Group 3, although the movement of medians shows a mildly higher level in t+2 for SFA episodes than for NSFA ones, we cannot conclude that monetary conditions are influencing the results. In general our results suggest that there is no evidence in our sample that an accommodating monetary policy stance is supporting the expansionary fiscal adjustment outcomes.

2. Consumption, Investment, and Net Export

Analyzing the behavior of GDP by components (private consumption, investment, and net export) may shed further light on the non-Keynesian effect of fiscal adjustment. For example, as argued above, if total consumption is observed to suddenly increase following the start of a fiscal adjustment process, there may be a stronger case in favor of the consumption-expectations hypothesis.²⁴ On the other hand, if investment increases, then there may be a stronger case in favor of the supply-side hypothesis.

Unfortunately, we could not derive particularly interesting results from our analysis mainly due to data insufficiency. For example, there are only 11 (in Group 1) and 17 (in Group 3) SFA episodes that contain private consumption data, so we were unable to analyze whether consumption or investment dominates the non-Keynesian effect of fiscal adjustment. While we expect future research to address this question, here we summarize what we have learned from our analysis that may direct future work.²⁵

First, for Group 1, the *median* of consumption growths (i.e., the growth rates of consumption as a share of GDP) in t to t+3 seems to be higher in SFA episodes than in NSFA episodes. (For Group 3, a similar trend is observed but the gap between the medians of SFA and NSFA episodes is smaller.) However, the *mean* of consumption growth does not suggest clear evidence in favor of the consumption-expectations hypothesis. Second, for both Groups 1 and 3, successful episodes show relatively higher cumulative growth rates in investment share from t to t+5. (This trend is found more clear for Group 3.) The result may be interpreted in favor of the supply-side hypothesis although we would require further analysis for conclusive evidence. Third, the trade balance is improved especially in SFA episodes for both Groups 1 and 3.²⁶

²³ Higher levels of means in t+3 and t+4 are driven by an outlier. Since only 12 observations exist, the mean may not be representative.

²⁴ In a rational expectations model, the argument would be refined to reflect the behavior following on the credible announcement of the fiscal adjustment. However data constraints on announcements prevent us from a more robust assessment of this hypothesis.

²⁵ The detailed results are not reported in this paper. Detailed tables and figures available from authors upon request.

²⁶ The result suggests that there may be some link between fiscal adjustment and trade balance. A possible explanation is that a declining fiscal deficit removes upward pressure on interest rates as competing (i.e., government) demand for loanable funds falls. The lower interest rates could initially lead to reduced capital inflows operating through the effects of the uncovered interest rate parity condition, hence resulting in a depreciation of the exchange rate that induces an improvement in the net trade position. While the expansionary fiscal adjustment through the effect on GDP could lead to stronger imports, in the context of small open economies the conjecture would be that exchange rate effect operating on exports and imports would dominate any impact over the short term from income-induced, expanded imports.

Table 12 of the Money/Quasi Money Growth Rate (%)	
Movement	

ITEM				SUC	CESSFUL						NOT	SUCCESS	FUL			
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1: Hi	gh-Incom	e OECD E	conomies	[13.0]												
Mean	10.2	11.1	14.1	13.9	10.2	19.8	18.1	7.2	18.5	19.5	16.8	23.8	19.9	14.4	16.4	17.3
25%	6.6	5.6	6.8	7.8	5.5	6.1	5.3	3.9	9.5	10.6	8.4	6.5	5.8	4.9	4.2	3.9
Median	9.4	9.2	12.1	12.2	9.4	12.4	8.2	6.5	15.6	15.9	14.9	14.7	16.4	13.2	9.2	11.1
75%	12.7	15.4	19.4	17.8	14.3	16.6	16.2	10.3	28.2	24.9	19.1	32.9	24.6	18.6	21.5	20.4
Min	-0.7	2.0	2.5	2.3	1.4	1.8	1.3	1.3	-10.0	0.5	3.1	1.4	0.5	-0.9	-5.2	1.7
Мах	27.8	27.8	32.6	32.6	24.9	116.0	116.0	15.0	57.2	65.3	48.4	116.0	80.1	48.4	70.3	80.1
SD	6.9	7.5	9.5	8.6	7.0	30.8	31.4	4.3	13.8	14.7	11.3	27.1	19.5	11.3	19.1	19.5
#				12								23				
Group 3: Mi	iddle-Inco	me Econ	omies [4.	4.9] [med	dian: 17.8	8]										
Mean	43.1	42.3	32.1	31.8	30.2	29.3	22.1	18.9	28.0	31.8	31.9	32.3	28.7	37.0	36.3	41.3
25%	15.2	16.3	15.0	11.7	12.2	12.9	9.8	7.0	12.3	14.3	11.7	13.2	11.1	10.8	10.9	10.6
Median	20.6	23.0	20.8	23.3	23.5	17.2	15.7	14.3	19.9	21.5	20.1	21.1	17.8	17.9	16.9	17.8
75%	37.2	33.7	31.7	29.8	31.5	29.1	25.6	24.3	28.3	34.0	36.8	31.3	26.6	26.4	28.0	26.0
Min	-4.4	-6.8	3.3	-57.2	3.3	0.6	-4.4	-5.7	-57.2	-2.9	-57.2	-0.4	-14.0	-27.6	-8.7	-27.6
Мах	411.8	411.8	118.5	276.0	132.7	219.3	104.0	118.5	334.0	297.3	322.5	411.8	322.5	1511.1	1462.7	1511.1
SD	72.0	66.3	30.1	44.2	26.6	37.9	21.8	19.5	36.9	37.2	43.6	47.9	39.3	128.4	121.7	161.2
#				50								149				





Group 1: Mean (Left) and three Quartiles (Right)





Group 3: Mean (Left) and three Quartiles (Right)

*Two bold lines denote the median.

C. Role of Initial Fiscal Conditions on Macro Variables

If economies are characterized by differing degrees of fiscal imbalance, should we expect the impact of fiscal adjustment on these economies also to vary given different initial conditions? Perotti (1999, 1399), for example, argues that "in times of fiscal stress, shocks to government revenues and, especially, expenditure have very different effects on private consumption than in normal times." In what follows we try to confirm this observation. Table 13 reports the movements of "normalized" growth rate for different "fiscal groups" in each of income Groups 1 and 3: high versus low (cyclically adjusted) primary deficits in Part A, and high versus low government debts in Part B.

- Primary Deficits: For SFA episodes in Group 1, SFA episodes with worse initial conditions on primary deficit result in higher levels of means and medians in all periods from t to t+5. However, such a distinction is not clear for Group 3 except for mean and median in t. (Perhaps, the government debt is a better indicator of initial fiscal condition because it is a cumulated version of primary deficits over many periods.)
- (ii) Government Debt: During periods of SFA in t, t+1 and t+2, both Groups 1 and 3 show relatively higher levels of mean and median in episodes with worse initial conditions compared to the ones with better conditions. For example, the Group 3 mean values for normalized growth are 1.0% (t), 1.8% (t+1), and 0.6% (t+2) with worse initial conditions, compared to -0.5% (t), -1.6% (t+1), and -0.6% (t+2) with better initial conditions. This is consistent with Perotti's (1999) findings. That is, it seems that different initial fiscal conditions matter in determining whether a fiscal adjustment can be expansionary or not.

D. Probit Analysis

In this subsection, we undertake a probit analysis to see whether specific fiscal or macroeconomic variables before or during the fiscal adjustment actually affect the success probability. We consider (subsets of) the following variables based on our discussion so far: (i) initial level of primary deficit at t-1, (ii) initial level of government debt at t-1, (iii) initial decrease in primary expenditure from t-1 to t, (iv) interim decrease in primary expenditure from t to t+2, (v) initial increase in total revenue from t-1 to t, (vi) interim increase in total revenue, (v) decrease in subsidy (especially for Group 1) from t-1 to t+2, (vi) decrease in capital expenditure (especially for Group 3) from t-1 to t+2, (vii) time-(t-1) "normalized" growth rate at t-1, and (viii) time-t "normalized" growth rate at t. The probit results are summarized in Table 14.

The first model (Model 1) in Table 14 includes all the explanatory variables listed above.

However, since "(v) decrease in subsidy from t-1 to t+2" and "(vi) decrease in capital expenditure from t-1 to t+2" are parts of the sum of "(iii) initial decrease in primary expenditure from t-1 to t'' and "(iv) interim decrease in primary expenditure from t to t+2", the results on these variables reported in Model 1 may reflect some interactions between them. Hence, to examine the effects of these variables separately, Model 2A excludes (v) and (vi) while Model 3A excludes (iii) and (iv). Also, one may argue that "(viii) time-t "normalized" growth rate at t" cannot be interpreted

Deficit
Primary
Adjusted)
A.(Cyclically

Table 13 Movement of the "Normalized" GDP Growth Rate: Do Initial Conditions on Primary Deficit Matter?

ITEM				WOR	SE FISCAI	CONDITI	[0N*					BETTE	ER FISCAL	. CONDITI	NO	
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1						1		1					0			
Mean	-1.0	0.3	1.1	0.8	0.1	0.5	0.1	0.5	0.6	1.4	1.0	0.4	-0.8	0.0	0.0	0.2
Median	-1.7	0.4	1.6	1.0	0.0	0.2	0.0	0.3	0.1	1.7	1.3	0.6	0.1	0.1	-0.4	0.1
SD	2.4	2.0	1.2	1.1	1.8	2.0	1.2	1.5	1.6	1.4	2.3	1.7	2.9	2.0	1.7	1.5
ŧ					12								13			
Group 3																
Mean	-1.0	2.1	2.1	0.2	0.4	-0.7	-0.3	-1.0	0.4	0.8	0.4	0.8	0.0	1.1	0.0	0.4
Median	-0.2	0.9	1.1	-0.1	0.3	0.0	-1.2	-0.2	0.3	0.8	0.7	0.3	0.6	1.3	0.4	0.7
SD	5.7	6.9	5.6	4.3	5.0	3.7	3.2	4.1	3.0	2.8	5.4	7.1	3.3	3.2	4.8	4.3
#					26								27			
*This table re	ports the m	ovement o	f normalize	ed GDP gro	wth rate fc) and "het-	ul episodes. ter fiscal c	. Successfu	Lepisodes	for each g	roup are e	qually divi	ded into tv	vo groups it is 5 26	again, "woi %	rse fiscal
		רארווימיויא	aujusteal	pilliai y ac) מוות הרו	רכו ווזרמי י		דום לבי כ		- י קטוט	1 · · · · · · · · · · · · · · · · · · ·	י קאטוט וי	, 15 15 7.5	.0.	

+ 0.4+ 2

ITEM				WOR	SE FISCA	L CONDIT	ION*					BETTI	ER FISCAI	L CONDIT	NO	
	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5	t-2	t-1	t	t+1	t+2	t+3	t+4	t+5
Group 1	c	6	0	C	L C	ć	0	0	c c	, ,	c c	ć	, ,	c c	ć	, ,
Median	0.6	6.0	0.9	0.7	0.3	-0.2	0.0 -0.4	-0.8 -0.8	-0.7	1.7	0.8 0	6.0	0.0	0.4 0.7	-0.2	2.1 0.8
OS #	2.2	1.8	1.4	0.7	1.6 8	1.7	1.6	1.6	1.3	1.6	2.5	2.0	3.3	2.4	1.7	1.1
ING P.																
Group 3 Mean	0.4	-0.4	1.0	1.8	0.6	-0.2	-0.3	-0.9	0.5	1.9	-0.5	-1.6	-0.6	1.4	1.3	1.2
Median	0.2	0.0	0.6	0.4	0.4	0.4	-0.3	-0.3	0.5	1.2	-1.0	-1.1	-0.1	1.2	1.3	1.6
OS No.	3.4	5.0	4.9	7.3	4.8	2.9	3.3	3.2	4.9	5.1	6.0	4.7	4.1	3.8	4.3	3.9
∓ 79					20								20			
*Same tab	le, but use	governm	ent debt i	instead of	F primary	deficit. T	he cut-of	ff levels a	re 45.9%	and 40.5	% for Gro	ups 1 an	d 3, respi	ectively.		

		GROL	JP 1				GROU	JP 3		
NAME OF VARIABLE	MODEL 1	MODEL 2A	MODEL 2B	MODEL 3A	MODEL 3 B	MODEL 1	MODEL 2A	MODEL 2 B	MODEL 3A	MODEL 3 B
Initial Level of Primary Deficit at <i>t-1</i>	-2.58 (0.16)	-2.59 (0.16)	-2.70 (0.22)	-2.60 (0.23)	-2.78 (0.22)	1.23 (0.46)	1.12 (0.49)	1.34 (0.38)	1.98 (0.17)	1.95 (0.17)
Initial Level of Government Debt at <i>t-1</i>	0.42 (0.05)*	0.49 (0.02)*	0.45 (0.04)*	0.39 (0.10)*	0.34 (0.16)	0.52 (0.12)	0.47 (0.15)	0.47 (0.15)	0.58 (0.06)*	0.58 (0.06)*
Initial Decrease in Primary Expenditure from <i>t-1</i> to <i>t</i>	-2.64 (0.68)	-0.86 (0.82)	3.45 (0.34)			3.23 (0.41)	1.12 (0.68)	2.08 (0.44)		
Interim Decrease in Primary Expenditure from <i>t</i> to <i>t+2</i>	8.92 (0.18)	9.69 (0.00)*	8.68 (0.00)*			10.21 (0.01)*	8.84 (0.01)*	8.61 (0.01)*		
Initial Increase in Total Revenue from <i>t-1</i> to <i>t</i>	4.07 (0.49)	4.42 (0.45)	4.56 (0.49)	5.79 (0.37)	5.71 (0.38)	2.96 (0.36)	2.36 (0.45)	2.81 (0.39)	1.59 (0.54)	1.62 (0.54)
Interim Increase in Total Revenue from <i>t</i> to <i>t+2</i>	11.16 (0.01)*	10.37 (0.01)*	9.88 (0.02)*	8.08 (0.04)*	8.71 (0.03)*	10.25 (0.00)*	9.90 (0.00)*	9.82 (0.00)*	4.84 (0.02)*	4.78 (0.01)*
Decrease in Subsidy from <i>t-1</i> to <i>t+2</i>	3.53 (0.63)			9.90 (0.01)*	10.71 (0.00)*	-1.45 (0.66)				
Decrease in Capital Expenditure from <i>t-1</i> to <i>t+2</i>	-6.37 (0.60)					-3.09 (0.42)			4.00 (0.10)*	4.57 (0.07)*
Time-(<i>t-1</i>) Normalized Growth Rate at <i>t-1</i>	2.90 (0.40)	3.06 (0.41)	3.18 (0.41)	1.27 (0.73)	2.08 (0.58)	3.39 (0.20)	2.45 (0.32)	1.34 (0.56)	2.60 (0.25)	1.96 (0.35)
Time- <i>t</i> Normalized Growth Rate at <i>t</i>	7.38 (0.02)*	8.00 (0.01)*		5.05 (0.09)*		-2.94 (0.12)	-2.86 (0.13)		-1.49 (0.38)	
No. of Observations	64	64	64	64	64	61	61	61	61	61
Log Likelihood	-22.14	-22.95	-26.97	-25.19	-26.79	-19.95	-20.27	-21.55	-27.06	-27.40

TABLE 14 Probit Analysis

"*" means the variable is significant at 10% level.

Note: The upper number in each cell is the marginal change in probability, and the lower number (in parenthesis) is the p-value.

as an *independent* variable because it may be affected by fiscal adjustment itself. In this sense, Models 2B and 3B excludes (viii) from Models 2A and 3A. We apply 90% confidence interval. The interpretation for each group is summarized below.

- (i) Group 1: First of all, we observe that "(ii) initial level of government debt at t-1" significantly affects the success probability in four out of five models. This is not surprising in light of our previous discussion. Second, and perhaps more interesting, not only "(iv) interim decrease in primary expenditure from t to t+2," but also "(vi) interim increase in total revenue from t to t+2," is significant. We have seen that relatively small fraction of improvement in fiscal deficit is due to a revenue increase in Group 1, but still, a higher level of revenue seems to help to achieve fiscal consolidations. Third, "(v) decrease in subsidy from t-1 to t+2," is significant in Models 3A and 3B, which is not surprising based on our previous discussion. Fourth, while "(vii) time-(t-1) normalized growth rate" is not significant, "(viii) time-t normalized growth rate" does affect the success of fiscal adjustment significantly. (This is in line with our earlier discussion in support of synchronizing fiscal adjustment to an expansionary phase of the business cycle.)
- (ii) Group 3: The result is more or less similar to the case of Group 1. First, the initial debt level still seems to be important; it is significant in Models 3A and 3B and only marginally rejected in other models. Second, just as in Group 1, both expenditure decrease and revenue increase are significant. Third, "(vi) decrease in capital expenditure (especially for Group 3) from t-1 to t+2," is significant. Perhaps the most notable difference between Group 1 is that now neither "(vii) time-(t-1)" nor "(viii) time-t normalized growth rate" is significant. This result implies that synchronizing fiscal adjustment to an expansionary phase is less important than other factors in fiscal consolidation in EMEs.

VI. CONCLUSION

The analysis of cross-country fiscal consolidation presented above provides a set of results that may guide policymakers to better understand what constitutes effective measures leading to SFA. Both composition and timing of fiscal adjustment matter for SFA. However, whereas expenditure cutbacks drive SFA in MEs, providing some validation for non-Keynesian results, revenue-enhancing measures combined with expenditure cutbacks constitute the driving force in EMEs. In particular, MEs adjust primarily by decreasing subsidies on average, which account for one half of expenditure savings. On the revenue side, MEs reduce social security taxes during SFA. In contrast, EMEs tend to pursue measures to increase total revenue, especially income tax and nontax revenue. This is likely to reflect greater scope for increasing taxes in many EMEs compared to MEs, given the relatively lower tax yields and perhaps less leeway for reducing a lower level of government expenditure.

The results for EMEs raise a note of caution as on average slightly over one half of the decrease in primary expenditure in EMEs during SFA is traced to a decrease in capital expenditure. This result is in line with standard models of political business cycles as politicians may find it to be politically expedient to reduce public investment instead of cutting back on salaries, pensions, and even subsidies, even at the cost of sacrificing the country's long-term growth rate. On the importance of timing, the results point to a greater likelihood of SFA, stemming from weak initial conditions such as higher levels of primary deficit and government debt and consequently higher debt servicing interest payments. The deadweight loss that is freed does tend to contribute to increasing resolve on the part of the policymaker to see the adjustment through to a successful end. Improved macroeconomic conditions at the start of the adjustment phase do appear to affect whether an economy can achieve a SFA.

We also examine whether monetary policy through an easing of liquidity conditions may be affecting the results above. This is not supported by the findings as there is no evidence of excess liquidity in the system coinciding with periods of SFA.

Finally, the findings point to a slightly more sustainable period of economic growth following SFA (although results according to group are mixed). In parallel, the unemployment rate is decreasing over time following SFA episodes, which is therefore generally consistent with the behavior of GDP growth.

An attempt is made to reconcile the statistical findings with several of the theoretical variants in support of the non-Keynesian results. The results are inconclusive given the limitations in the GDP decomposition data. However, on the basis of the behavior of investment, there appears to be a weak inclination in favor of the supply-side hypothesis as compared to the consumption– expectation hypothesis.

Finally, we reviewed the results derived above by undertaking a probit analysis based on various fiscal and macroeconomic variables. The probit findings indicate that initial fiscal conditions matter; both the revenue and expenditure sides are significant and important (although on average, expenditure side has larger contribution, especially in Group 1); subsidy component or capital expenditure component is significant in each group; and finally, the time-*t* growth rate is significant for MEs.

While our findings are based on relatively simple statistical analysis, we expect future research to provide more statistically robust results. For example, future research may control for the effects of country or period-specific characteristics instead of depending on basic comparisons of averages or medians or simple probit analyses. Furthermore, with a richer dataset it may be possible to provide more conclusive evidence as to which one of the two hypotheses—consumption–expectation or supply-side—better explains the non-Keynesian results of fiscal adjustment. Finally, a richer data set may lead to a better understanding of variables such as political orientation of the government that may have a strong bearing on the credibility of announced fiscal reforms.

APPENDIX

A. Data

Most of the fiscal and macroeconomic data—government revenue, government expenditure, government debt, real GDP growth, unemployment rate, and many others—used in this paper are obtained from *World Development Indicators 2004* by the World Bank. For each of these variables, we have the observations of 33 years (from 1970 to 2002) for 208 economies, although many "missing" observations exist especially for EMEs.

To examine how variables react to fiscal adjustment in different income groups, we classify those 208 economies into the following four groups: (i) Group 1: high-income OECD economies; (ii) Group 2: high-income non-OECD economies, (with 2002 income per-capita of US\$9,075 and above); (iii) Group 3: middle-income economies (with 2002 income per capita between US\$735 and US\$9,075,); and (iv) Group 4: low-income economies (with income per capita of US\$735 and below). The cutoff levels of \$9,075 and \$735 are adapted from the World Bank criterion. High-income economies are regrouped into OECD and non-OECD economies for the following reasons: First, OECD economies are largely open economies and therefore require (and are characterized by) a relatively higher degree of fiscal discipline than non-OECD ones. Second, there are many oil-exporting (high-income) economies among the high-income non-OECD group that may be subject to different country risk assessments from more conventional OECD countries as oil reserves may reduce solvency constraints.

Throughout this paper, we present the results for Group 1 and Group 3 only. Group 2 results were excluded for the purpose of easy reading as the results were in line with findings straddling Group1 and Group 3 results. Group 4, or low-income economies, is excluded from our analysis because too many observations are missing, and even for existing data, reliability may be an issue.

B. Cyclical Adjustment

The performance of fiscal variables depends on the business cycle: Automatic stabilizer properties suggest that the government *revenue* adjusts quickly to the business cycle, while the government *expenditure* does not. Hence, not adjusting the data may lead to distortions in the results and in particular in the (unadjusted) government deficit as changes in the deficit may simply reflect the effects of the cyclical change. A conventional way to deal with this problem is to "cyclically adjust" the fiscal data, following the method suggested by Blanchard (1990b).

Blanchard's method uses the unemployment rate to reflect the changes in the business cycle: The "cyclically adjusted fiscal data" are obtained by assuming the unemployment rate had been the same as in the previous year. First, we run the following regression for each of the economies:

fiscal
$$_{t} = \alpha + \beta \cdot u_{t} + \varepsilon_{t}$$

where *fiscal* $_t$ is the fiscal variable of our interest (at time t), α is the intercept, β is the slope, u_t is the unemployment rate, and ε_t is the error term. For a and b (estimates for α and β), the cyclically adjusted fiscal variable for time t is obtained by calculating:

$$adjusted_fiscal_t = fiscal_t - b (u_t - u_{t-1}).$$

However, in view of our large sample size comprising MEs and EMEs, the unemployment rates for EMEs (including middle-income economies) is problematic as data are often missing or inaccurate. Under the circumstances, we instead use the "GDP growth rate *net of the GDP trend*" (where the GDP trend is calculated using the Hodrick-Prescott filter).¹

Government expenditure, government revenue, and primary deficit are cyclically adjusted using this variable instead of the unemployment rate in the equations above. In addition, to avoid the unnecessarily large impacts of sizable changes to GDP growth rate (for example, during and after a crisis) on cyclically adjusted fiscal variables, we have deleted from our dataset those cyclically adjusted government deficit or revenue observations that lie outside a range of between 10% and 70% of GDP.²

APPENDIX TABLE 1 LIST OF EPISODES OF FISCAL ADJUSTMENT

For each episode of fiscal adjustment, this table lists the country name, the country's income group (IG), and the year of the episode. The IG is defined in the Data section of the Appendix (see above). For example, "1" denotes that the country is a high-income OECD economy (with income per capita above US\$9,075).

¹ A more detailed explanation on the "GDP growth rate net of the GDP trend" is provided in Section IV.

² As in Alesina, Perotti, and Tavares (1998), we also find that there is not much difference to the results whether we use "cyclically adjusted" or "unadjusted" fiscal variables. In this paper, we present the results based on "cyclically adjusted" fiscal variables.

Successful Fiscal Adjustments (110 episodes)

amas, The (2) 1977 rain (2) 1977 arus (2) 1989 arus (3) 1994 arus (3) 1994 gium (1) 1984 ttan (4) 1991 ttan (4) 1991 ttan (4) 1991 tvia (3) 1995 swana (3) 1983 swana (3) 1983								
(2) 1989 (3) 1994 (3) 1994 (3) 1994 (3) 1994 (4) 1998 (4) 1998 (3) 1994 (4) 1995 (3) 1995 (3) 1995 (3) 1984 (3) 1983 (3) 1983 (3) 1983 (3) 1983 (3) 1984 (4) 1983 (3) 1984 (4) 1983 (4)	Eavot. Arab Rep.	(3) 197	7 Liberia	(4)	1983	Peru	(3)	995
n (3) 1994 (3) 1997 (3) 1997 (3) 1998 (4) 1984 (4) 1991 (4) 1991 (3) 1994 (3) 1995 na (3) 1983 (3) 1983 (3) 1983	Egypt, Arab Rep.	(3) 195	1 Luxembourg	(1)	1973	Peru	(3) 1	966
(3) 1997 (3) 1997 (3) 1998 (4) 1984 (4) 1991 (4) 1994 (3) 1994 (3) 1995 (3) 1995 (3) 1983 (3) 1983 (3) 1983 (3) 1983 (3) 1983	Finland	(1) 195	5 Luxembourg	(1)	1983	Singapore	(2) 1	988
(3) 1998 (1) 1984 (4) 1984 (4) 1984 (4) 1991 (4) 1991 (4) 1991 (3) 1995 (3) 1995 (3) 1983 (3) 1983 (3) 1983 (3) 1984 (3)	Greece	(1) 199	0 Malawi	(4)	1981	Solomon Islands ((4) 1	988
m (1) 1984 n (4) 1989 n (4) 1991 n (4) 1991 n (3) 1994 n (3) 1995 n (3) 1985 n (3) 1983 n (3) 1983 n (3) 1984	Greece	(1) 199	1 Malawi	(4)	1983	South Africa ((3) 1	994
n (4) 1989 n (4) 1981 a (3) 1994 ana (3) 1995 ana (3) 1977 ana (3) 1983 ana (3) 1983	Hungary	(3) 199	4 Malaysia	(3)	1982	Sri Lanka ((3) 1	981
n (4) 1991 a (3) 1994 ana (3) 1995 ana (3) 1977 ana (3) 1983 ana (3) 1984	Hungary	(3) 199	5 Malaysia	(3)	1983	Sri Lanka ((3) 1	989
a (3) 1994 a (3) 1995 ana (3) 1977 ana (3) 1984 ana (3) 1984	Iceland	(1) 199	5 Malta	(2)	1975	St. Kitts and Nevis ((3) 1	989
a (3) 1995 ana (3) 1977 ana (3) 1983 ana (3) 1984 ana (3) 1984	India	(4) 199	3 Mauritius	(3)	1983	Swaziland ((3) 1	975
ana (3) 1977 ana (3) 1983 ana (3) 1984 ana (3) 1984	Indonesia	(4) 198	7 Mauritius	(3)	1983	Swaziland ((3) 1	987
ana (3) 1983 ana (3) 1984 	Iran, Islamic Rep.	(3) 198	3 Mexico	(3)	1995	Sweden ((1) 1	984
ana (3) 1984 J: (7) 1002	Iran, Islamic Rep.	(3) 195	0 Moldova	(4)	6661	Sweden ((1) 1	994
	Iran, Islamic Rep.	(3) 199	1 Morocco	(3)	1982	Sweden ((1) 1	995
01 (4) TAA2	Israel	(2) 198	5 Morocco	(3)	1988	Thailand ((3) 1	988
a (1) 1996	Israel	(2) 198	6 Morocco	(3)	0661	Thailand ((3) 1	989
a (1) 1997	Israel	(2) 199	3 Namibia	(3)	1988	Togo ((4) 1	982
(3) 1977	Israel	(2) 199	7 New Zealand	(1)	1984	Togo ((4) 1	983
(3) 1978	Italy	(1) 198	8 New Zealand	(1)	1985	Tunisia	(3) 1	979
(3) 1990	Jamaica	(3) 195	7 New Zealand	(1)	1995	Turkey ((3) 1	997
(3) 1993	Jamaica	(3) 199	8 Norway	(1)	1980	United Kingdom ((1) 1	988
, Rep. (4) 1994	Jordan	(3) 198	3 Norway	(1)	1981	United Kingdom ((1) 1	994
Rica (3) 1981	Jordan	(3) 199	0 Norway	(1)	1995	United Kingdom ((1) 1	995
('Ivoire (4) 1996	Jordan	(3) 199	2 Oman	(3)	1976	Uruguay ((3) 1	976
('Ivoire (4) 1997	Kenya	(4) 199	0 Oman	(3)	1977	Uruguay ((3) 1	985
а (3) 1995	Kuwait	(2) 199	4 Oman	(3)	1987	Uruguay ((3) 1	060
rk (1) 1983	Kuwait	(2) 199	5 Oman	(3)	1995	Venezuela, RB ((3) 1	973
irk (1) 1984	Kuwait	(2) 199	6 Papua New Guinea	(4)	1983	Yemen, Rep. ((4) 1	995
ark (1) 1985	Lesotho	(4) 198	6					

continued.

Not Successfi	ul Fis	cal Adjı	ustment (345 ef	oisodu	es)						
COUNTRY NAME	IG	YEAR	COUNTRY NAME	JI	YEAR	COUNTRY NAME	IJG	YEAR	COUNTRY NAME	IG	YEAR
Algeria	(3)	1996	Estonia	(3)	1998	Lesotho	(4)	1974	Romania	(3)	1997
Argentina	(3)	1998	Ethiopia	(4)	1984	Liberia	(4)	1980	Seychelles	(3)	1994
Australia	(1)	1980	Ethiopia	(4)	1988	Liberia	(4)	1984	Sierra Leone	(4)	1976
Australia	(1)	1983	Fiji	(3)	1974	Luxembourg	(1)	1974	Singapore	(2)	1973
Australia	(1)	1987	Fiji	(3)	1978	Luxembourg	(1)	1977	Singapore	(2)	1974
Austria	(1)	1977	Fiji	(3)	1980	Luxembourg	(1)	1985	Singapore	(2)	1977
Austria	(1)	1980	Fiji	(3)	1982	Luxembourg	(1)	1987	Singapore	(2)	1982
Austria	(1)	1984	Fiji	(3)	1985	Luxembourg	(1)	1989	Singapore	(2)	1985
Bahamas, The	(2)	1983	Fiji	(3)	1987	Luxembourg	(1)	1992	Singapore	(2)	1989
Bahamas, The	(2)	1990	Fiji	(3)	1990	Luxembourg	(1)	1994	Singapore	(2)	1994
Bahamas, The	(2)	1991	Fiji	(3)	1993	Macao, China	(2)	1998	Singapore	(2)	1998
Bahamas, The	(2)	1994	Finland	(1)	1976	Madagascar	(4)	1990	Solomon Islands	(4)	1979
Bahamas, The	(2)	1997	Finland	(1)	1980	Malawi	(4)	1976	Solomon Islands	(4)	1981
Bahrain	(2)	1985	Finland	(1)	1984	Malawi	(4)	1984	Solomon Islands	(4)	1983
Bahrain	(2)	1987	Finland	(1)	1988	Malaysia	(3)	1978	Solomon Islands	(4)	1984
Bahrain	(2)	1990	France	(1)	1977	Malaysia	(3)	1979	South Africa	(3)	1975
Bahrain	(2)	1993	France	(1)	1989	Malaysia	(3)	1984	South Africa	(3)	1979
Bahrain	(2)	1995	Gambia, The	(4)	1976	Malaysia	(3)	1985	South Africa	(3)	1982
Bahrain	(2)	1996	Gambia, The	(4)	1979	Malaysia	(3)	1988	South Africa	(3)	1985
Barbados	(2)	1974	Germany	(1)	1977	Malaysia	(3)	1991	South Africa	(3)	1989
Barbados	(2)	1976	Germany	(1)	1980	Malaysia	(3)	1994	South Africa	(3)	1995
Barbados	(2)	1978	Germany	(1)	1989	Maldives	(3)	1998	South Africa	(3)	1997
Barbados	(2)	1980	Germany	(1)	1995	Malta	(2)	1978	Spain	(1)	1973
Barbados	(2)	1984	Ghana	(4)	1986	Malta	(2)	1988	Spain	(1)	1986
Barbados	(2)	1985	Greece	(1)	1983	Malta	(2)	1992	Sri Lanka	(3)	1977
Belgium	(1)	1974	Greece	(1)	1987	Malta	(2)	1995	Sri Lanka	(3)	1983

continued.

COUNTRY NAME	IG	YEAR	COUNTRY NAME	IG	YEAR	COUNTRY NAME	IG	YEAR	COUNTRY NAME	IG	YEAR
Belgium	(1)	1982	Greece	(1)	1992	Mauritius	(3)	1997	Sri Lanka	(3)	1984
Belgium	(1)	1985	Greece	(1)	1994	Mexico	(3)	1980	Sri Lanka	(3)	1990
Bhutan	(4)	1993	Guyana	(3)	1977	Mexico	(3)	1983	Sri Lanka	(3)	1992
Bhutan	(4)	1997	Guyana	(3)	1978	Mongolia	(4)	1995	St. Kitts and Nevis	(3)	1990
Bhutan	(4)	1998	Guyana	(3)	1981	Mongolia	(4)	1996	Swaziland	(3)	1974
Botswana	(3)	1979	Hungary	(3)	1984	Morocco	(3)	1973	Swaziland	(3)	1979
Botswana	(3)	1985	Hungary	(3)	1988	Morocco	(3)	1978	Swaziland	(3)	1982
Botswana	(3)	1988	Hungary	(3)	1990	Morocco	(3)	1979	Sweden	(1)	1976
Botswana	(3)	1990	Iceland	(1)	1975	Morocco	(3)	1984	Sweden	(1)	1985
Brazil	(3)	1983	Iceland	(1)	1978	Namibia	(3)	1989	Sweden	(1)	1987
Brazil	(3)	1985	Iceland	(1)	1981	Namibia	(3)	1994	Sweden	(1)	1996
Bulgaria	(3)	1991	Iceland	(1)	1988	Namibia	(3)	1997	Switzerland	(1)	1978
Bulgaria	(3)	1994	Iceland	(1)	1992	Netherlands	(1)	1977	Switzerland	(1)	1981
Burundi	(4)	1994	India	(4)	1978	Netherlands	(1)	1985	Thailand	(3)	1977
Cameroon	(4)	1979	India	(4)	1981	Netherlands	(1)	1991	Thailand	(3)	1979
Cameroon	(4)	1982	India	(4)	1982	New Zealand	(1)	1976	Thailand	(3)	1983
Cameroon	(4)	1988	India	(4)	1984	New Zealand	(1)	1979	Tunisia	(3)	1975
Cameroon	(4)	1990	India	(4)	1989	New Zealand	(1)	1987	Tunisia	(3)	1981
Cameroon	(4)	1992	India	(4)	1991	Nicaragua	(4)	1973	Tunisia	(3)	1984
Chile	(3)	1975	India	(4)	1995	Nicaragua	(4)	1978	Tunisia	(3)	1986
Chile	(3)	1979	Indonesia	(4)	1974	Nicaragua	(4)	1997	Tunisia	(3)	1988
Chile	(3)	1980	Indonesia	(4)	1979	Nicaragua	(4)	1998	Tunisia	(3)	1991
Chile	(3)	1985	Indonesia	(4)	1982	Norway	(1)	1985	Tunisia	(3)	1993
Chile	(3)	1996	Indonesia	(4)	1984	Norway	(1)	1996	Tunisia	(3)	1997
Colombia	(3)	1985	Indonesia	(4)	1990	Oman	(3)	1979	Turkey	(3)	1974
Colombia	(3)	1987	Indonesia	(4)	1994	Oman	(3)	1981	Turkey	(3)	1978
Colombia	(3)	1993	Iran, Islamic Rep.	(3)	1977	Oman	(3)	1989	Turkey	(3)	1981
Congo, Dem. Rep.	(4)	1975	Iran	(3)	1984	Oman	(3)	1990	Turkey	(3)	1986
Congo, Dem. Rep.	(4)	1989	Iran	(3)	1986	Oman	(3)	1996	Turkey	(3)	1990
Congo, Rep.	(4)	1995	Iran	(3)	1988	Oman	(3)	1997	Turkey	(3)	1992
Congo, Rep.	(4)	1996	Iran	(3)	1992	Pakistan	(4)	1980	Turkey	(3)	1995
Costa Rica	(3)	1982	Iran	(3)	1997	Pakistan	(4)	1982	Turkey	(3)	1998
Costa Rica	(3)	1984	Ireland	(1)	1976	Pakistan	(4)	1990	United Kingdom	(1)	1980
Costa Rica	(3)	1988	Ireland	(1)	1983	Pakistan	(4)	1998	United Kingdom	(1)	1982
Costa Rica	(3)	1991	Ireland	(1)	1988	Panama	(3)	1977	United States	(1)	1987
Costa Rica	(3)	1992	Ireland	(1)	1989	Panama	(3)	1980	Uruguay	(3)	1979
Costa Rica	(3)	1995	Israel	(2)	1974	Panama	(3)	1983	Uruguay	(3)	1981
Cote d'Ivoire	(4)	1986	Israel	(2)	1977	Panama	(3)	1985	Uruguay	(3)	1987
Croatia	(3)	1993	Israel	(2)	1978	Panama	(3)	1990	Uruguay	(3)	1988
Croatia	(3)	1996	Israel	(2)	1982	Panama	(3)	1991	Uruguay	(3)	1993
Croatia	(3)	1998	Italy	(1)	1976	Papua New Guinea	(4)	1980	Uruguay	(3)	1995
Cyprus	(2)	1984	Italy	(1)	1993	Papua New Guinea	(4)	1984	Uruguay	(3)	1997
Cyprus	(2)	1986	Italy	(1)	1995	Papua New Guinea	(4)	1987	Vanuatu	(3)	1984
Cyprus	(2)	1989	Jordan	(3)	1981	Papua New Guinea	(4)	1995	Vanuatu	(3)	1996
Cyprus	(2)	1993	Jordan	(3)	1984	Papua New Guinea	(4)	1996	Venezuela, RB	(3)	1974
Czech Republic	(3)	1996	Jordan	(3)	1986	Paraguay	(3)	1994	Venezuela	(3)	1979
Denmark	(1)	1977	Kenya	(4)	1974	Paraguay	(3)	1996	Venezuela	(3)	1981
Denmark	(1)	1986	Kenya	(4)	1978	Paraguay	(3)	1998	Venezuela	(3)	1984
Denmark	(1)	1996	Kenya	(4)	1983	Peru	(3)	1975	Venezuela	(3)	1989
Dominican Rep.	(3)	1975	Kenya	(4)	1986	Peru	(3)	1998	Venezuela, RB	(3)	1990
Dominican Rep.	(3)	1980	Kenya	(4)	1991	Philippines	(3)	1979	Venezuela, RB	(3)	1996
Egypt. Arab Rep.	(3)	1978	Kenya	(4)	1994	Poland	(3)	1996	Vietnam	(4)	1998
Egypt. Arab Rep.	(3)	1980	Korea, Rep.	(1)	1977	Portugal	(1)	1983	Yemen, Rep.	(4)	1996
Egypt. Arab Rep.	(3)	1981	Korea	(1)	1984	Portugal	(1)	1986	Zambia	(4)	1974
Egypt. Arab Rep.	(3)	1983	Korea	(4)	1987	Portugal	(1)	1988	Zambia	(4)	1977
Egypt. Arab Rep.	(3)	1987	Korea	(4)	1992	Romania	(3)	1984	Zambia	(4)	1978
Egypt. Arab Rep.	(3)	1989	Latvia	(3)	1998	Komania	(3)	1987	Zimbabwe	(4)	1981
Egypt. Arab Rep.	(3)	1993	Lebanon	(3)	1995	Romania	(3)	1989	Zimbabwe	(4)	1986
Egypt. Arab Rep.	(3)	1994	Lesotho	(4)	19/3	Komania	(3)	1993	Zimbabwe	(4)	1989
Estonia	(3)	1996									

ITEM	SUSTAINED: 84	NOT SUSTAINED: 295	UNCLASSIFIED: 76
Successful: 110	52	35	23
Not Successful: 345	32	260	53

Appendix Table 2 Breakdown of Fiscal Adjustment

REFERENCES

- Alesina, A., and S. Ardagna. 1998. "Fiscal Adjustment: Why They Can Be Expansionary." *Economic Policy* 13(27):488-545.
- Alesina, A., S. Ardagna, R. Perotti, and F. Schiantarelli. 2002. "Fiscal Policy, Profits, and Investment." *American Economic Review* 92(3):571-89.
- Alesina, A., and R. Perotti. 1995. "Fiscal Expansions and Adjustments in OECD Countries." *Economic Policy* 10(2):205-48.
- Alesina, A., and R. Perotti. 1997. "Fiscal Adjustment in OECD Countries: Composition and Macroeconomic Effects." International Monetary Fund Staff Papers 44:210-248.
- Alesina, A., R. Perotti, and J. Tavares. 1998. "The Political Economy of Fiscal Adjustments." *Brookings Papers* on Economic Activity 1(Spring):197-266.
- Ardagna, S. 2004. "Fiscal Stabilizations: When do They Work and Why?" *European Economic Review* 48(5):1047-74.
- Bertola, G., and A. Drazen. 1993. "Trigger Points and Budget Cuts: Explaining the Effects of Fiscal Austerity." American Economic Review 83(1):11-26.
- Blanchard, O. 1990a. "Comment." NBER Macroeconomics Annual 5:111-6.
- Blanchard, O. 1990b. Suggestion for a New Set of Fiscal Indicators. OECD Working Paper No. 79, Organisation for Economic Cooperation, Paris.

 Calderon, C., W. Easterly, and L. Serven. 2002a. How did Latin America's Infrastructure Fare in the Era of Macroeconomic Crises? Central Bank of Chile Working Paper No. 185, Banco Central de Chile, Santiago.
 2002b. "Infrastructure Compression and Public Sector Solvency in Latin America. Central Bank of Chile

- Working Paper No. 187, Banco Central de Chile, Santiago.
- Giavazzi, F., and M. Pagano. 1996. "Non-Keynesian Effects of Fiscal Policy Changes: International Evidence and the Swedish Experience." *Swedish Economic Policy Review* 3(Spring):135-65.
- IMF. 2004. Public Investment and Fiscal Policy. International Monetary Fund, Washington, DC.
- Obstfeld, M. 1998. "Open-Economy Macroeconomics: Developments in Theory and Policy." Scandinavian Journal of Economics 101(1):247-75.
- OECD. 2003. "Fiscal Stance Over the Cycle." Economic Outlook. Paris.

Perotti, R. 1999. "Fiscal Policy in Good Times and Bad." Quarterly Journal of Economics 114(4):1399-436.

Turnovsky, S. 2000. Methods of Macroeconomic Dynamics. 2nd ed. Cambridge, MA: MIT Press.

World Bank. 2004. World Development Indicators 2004. Washington, DC.

PUBLICATIONS FROM THE ECONOMICS AND RESEARCH DEPARTMENT

ERD WORKING PAPER SERIES (WPS)

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

No.	1	Capitalizing on Globalization —Barry Eichengreen January 2002
No.	2	Policy-based Lending and Poverty Reduction:
		and Options
		-Richard Bolt and Manabu Fujimura, January
No.	3	The Automotive Supply Chain: Global Trends
		and Asian Perspectives
No.	4	—Francisco Veloso and Rajiv Kumar, January 2002 International Competitiveneess of Asian Firms: An Analytical Framework
		-Rajiv Kumar and Doren Chadee, February 2002
No.	5	The International Competitiveness of Asian
		Economies in the Apparel Commodity Chain
No.	6	Monetary and Financial Cooperation in East
		Asia—The Chiang Mai Initiative and Beyond
No	7	-Pradumna B. Rana, February 2002
10.	1	Inequality and Growth in the Philippines
		-Arsenio M. Balisacan and Ernesto M. Pernia,
	0	March 2002
No.	8	-Anil B Declalibar April 2002
No.	9	Microfinance in Northeast Thailand: Who Benefits
		and How Much?
No	10	-Brett E. Coleman, April 2002
10.	10	Developing Asia
		Raghav Gaiha. Ernesto M. Pernia. Marv Racelis
		with the assistance of Marita Concepcion Castro-
		Guevara, Liza L. Lim, Pilipinas F. Quising, May
No.	11	The European Social Model: Lessons for
		Developing Countries
	10	-Assar Lindbeck, May 2002
No.	12	Costs and Benefits of a Common Currency for ASEAN
		—Srinivasa Madhur, May 2002
No.	13	Monetary Cooperation in East Asia: A Survey
No.	14	Toward A Political Economy Approach
		to Policy-based Lending
NT	15	-George Abonyi, May 2002
No.	15	A Framework for Establishing Priorities in a Country Poverty Reduction Strategy
		-Ron Duncan and Steve Pollard, June 2002
No.	16	The Role of Infrastructure in Land-use Dynamics
		and Rice Production in Viet Nam's Mekong River
		-Christopher Edmonds, July 2002
No.	17	Effect of Decentralization Strategy on
		Macroeconomic Stability in Thailand
No.	18	Poverty and Patterns of Growth
		-Rana Hasan and M. G. Quibria, August 2002
No.	19	Why are Some Countries Richer than Others? A Reassessment of Mankiw-Romer-Weil's Test of

the Neoclassical Growth Model

- -Jesus Felipe and John McCombie, August 2002 No. 20 Modernization and Son Preference in People's Republic of China -Robin Burgess and Juzhong Zhuang, September 2002
- No. 21 The Doha Agenda and Development: A View from the Uruguay Round
- -J. Michael Finger, September 2002 No. 22 Conceptual Issues in the Role of Education Decentralization in Promoting Effective Schooling in Asian Developing Countries -Jere R. Behrman, Anil B. Deolalikar, and Lee-Ying Son, September 2002
- No. 23 Promoting Effective Schooling through Education Decentralization in Bangladesh, Indonesia, and Philippines
 —Jere R. Behrman, Anil B. Deolalikar, and Lee-Ying Son, September 2002
- No. 24 Financial Opening under the WTO Agreement in Selected Asian Countries: Progress and Issues —Yun-Hwan Kim, September 2002
- No. 25 Revisiting Growth and Poverty Reduction in Indonesia: What Do Subnational Data Show? —Arsenio M. Balisacan, Ernesto M. Pernia, and Abuzar Asra, October 2002
- No. 26 Causes of the 1997 Asian Financial Crisis: What Can an Early Warning System Model Tell Us? —Juzhong Zhuang and J. Malcolm Dowling, October 2002
- No. 27 Digital Divide: Determinants and Policies with Special Reference to Asia —M. G. Quibria, Shamsun N. Ahmed, Ted Tschang, and Mari-Len Reyes-Macasaquit, October 2002
- No. 28 Regional Cooperation in Asia: Long-term Progress, Recent Retrogression, and the Way Forward —Ramgopal Agarwala and Brahm Prakash, October 2002
- No. 29 How can Cambodia, Lao PDR, Myanmar, and Viet Nam Cope with Revenue Lost Due to AFTA Tariff Reductions?
- —Kanokpan Lao-Araya, November 2002
 No. 30 Asian Regionalism and Its Effects on Trade in the 1980s and 1990s
 —Ramon Clarete, Christopher Edmonds, and Jessica Seddon Wallack, November 2002
- No. 31 New Economy and the Effects of Industrial Structures on International Equity Market Correlations --Cyn-Young Park and Jaejoon Woo, December 2002
- No. 32 Leading Indicators of Business Cycles in Malaysia and the Philippines —Wenda Zhang and Juzhong Zhuang, December 2002
- No. 33 Technological Spillovers from Foreign Direct Investment—A Survey —Emma Xiaoqin Fan, December 2002

No.	34	Economic Openness and Regional Development in the Philippines
No	35	-Ernesto M. Pernia and Pilipinas F. Quising, January 2003 Bond Market Development in East Asia:
1101		Issues and Challenges —Raul Fabella and Srinivasa Madhur, January
No.	36	2003 Environment Statistics in Central Asia: Progress and Prospects
		-Robert Ballance and Bishnu D. Pant, March 2003
No.	37	Electricity Demand in the People's Republic of China: Investment Requirement and Environmental Impact
No.	38	-Bo Q. Lin, March 2003 Foreign Direct Investment in Developing Asia: Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations
		-Douglas H. Brooks, Emma Xiaoqin Fan, and Lea R. Sumulong, April 2003
No.	39	The Political Economy of Good Governance for Poverty Alleviation Policies
No.	40	Narayan Lakshman, April 2003 The Puzzle of Social Capital A Critical Review
No.	41	-M. G. Quibria, May 2003 Industrial Structure, Technical Change, and the Role of Government in Development of the Electronics and Information Industry in
NT	40	Taipei,China —Yeo Lin, May 2003
No.	42	in Viet Nam —Arsenio M. Balisacan, Ernesto M. Pernia, and
No.	43	Gemma Esther B. Estrada, June 2003 Why Has Income Inequality in Thailand Increased? An Analysis Using 1975-1998 Surveys
No.	44	— <i>Taizo Motonishi, June 2003</i> Welfare Impacts of Electricity Generation Sector Reform in the Philippines
No.	45	Natsuko Toba, June 2003 A Review of Commitment Savings Products in Developing Countries Nava Ashrof Nathalia Cone, Dean S. Karlan
No.	46	and Wesley Yin, July 2003 Local Government Finance, Private Resources, and Local Credit Markets in Asia —Roberto de Vera and Yun-Hwan Kim, October
No.	47	2003 Excess Investment and Efficiency Loss During Reforms: The Case of Provincial-level Fixed-Asset Investment in People's Republic of China
No.	48	-Duo Qin and Haiyan Song, October 2003 Is Export-led Growth Passe? Implications for Developing Asia
No.	49	-Jesus Felipe, December 2003 Changing Bank Lending Behavior and Corporate Financing in Asia-Some Research Issues -Emma Xiaoqin Fan and Akiko Terada-Hagiwara
No.	50	Is People's Republic of China's Rising Services Sector Leading to Cost Disease?
No.	51	Poverty Estimates in India: Some Key Issues
No.	52	Restructuring and Regulatory Reform in the Power Sector: Review of Experience and Issues
No.	53	Peter Choynowski, May 2004 Competitiveness, Income Distribution, and Growth in the Philippines: What Does the Long-run Fuidonce Show?
		-Jesus Felipe and Grace C. Sipin, June 2004

No. 54	Practices of Poverty Measurement and Poverty
	Profile of Bangladesh
	—Faizuddin Ahmed, August 2004

- No. 55 Experience of Asian Asset Management Companies: Do They Increase Moral Hazard? —Evidence from Thailand —Akiko Terada-Hagiwara and Gloria Pasadilla, September 2004
- No. 56 Viet Nam: Foreign Direct Investment and Postcrisis Regional Integration —Vittorio Leproux and Douglas H. Brooks, September 2004
- No. 57 Practices of Poverty Measurement and Poverty Profile of Nepal
- -Devendra Chhetry, September 2004 No. 58 Monetary Poverty Estimates in Sri Lanka: Selected Issues -Neranjana Gunetilleke and Dinushka Senanayake, October 2004
- No. 59 Labor Market Distortions, Rural-Urban Inequality, and the Opening of People's Republic of China's Economy
- -Thomas Hertel and Fan Zhai, November 2004 No. 60 Measuring Competitiveness in the World's Smallest
- Economies: Introducing the SSMECI —Ganeshan Wignaraja and David Joiner, November 2004
- No. 61 Foreign Exchange Reserves, Exchange Rate Regimes, and Monetary Policy: Issues in Asia —Akiko Terada-Hagiwara, January 2005
- No. 62 A Small Macroeconometric Model of the Philippine Economy

 Geoffrey Ducanes, Marie Anne Cagas, Duo Qin, Pilipinas Quising, and Nedelyn Magtibay-Ramos, January 2005
- No. 63 Developing the Market for Local Currency Bonds by Foreign Issuers: Lessons from Asia —Tobias Hoschka, February 2005
- No. 64 Empirical Assessment of Sustainability and Feasibility of Government Debt: The Philippines Case
 —Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, and Pilipinas Quising,
- February 2005 No. 65 Poverty and Foreign Aid Evidence from Cross-Country Data —Abuzar Asra, Gemma Estrada, Yangseom Kim, and M. G. Quibria, March 2005
- No. 66 Measuring Efficiency of Macro Systems: An Application to Millennium Development Goal Attainment —Ajay Tandon, March 2005
- No. 67 Banks and Corporate Debt Market Development —Paul Dickie and Emma Xiaogin Fan, April 2005
- No. 68 Local Currency Financing—The Next Frontier for MDBs?
- —Tobias C. Hoschka, April 2005
 No. 69 Export or Domestic-Led Growth in Asia?
 —Jesus Felipe and Joseph Lim, May 2005
- No. 70 Policy Reform in Viet Nam and the Asian Development Bank's State-owned Enterprise Reform and Corporate Governance Program Loan —George Abonyi, August 2005
- No. 71 Policy Reform in Thailand and the Asian Development Bank's Agricultural Sector Program Loan —George Abonyi, September 2005
- No. 72 Can the Poor Benefit from the Doha Agenda? The Case of Indonesia —Douglas H. Brooks and Guntur Sugiyarto, October 2005
- No. 73 Impacts of the Doha Development Agenda on People's Republic of China: The Role of Complementary Education Reforms —Fan Zhai and Thomas Hertel, October 2005

No.	74	Growth and Trade Horizons for Asia: Long-term
		Forecasts for Regional Integration
		-David Roland-Holst, Jean-Pierre Verbiest, and
		Fan Zhai, November 2005
No.	75	Macroeconomic Impact of HIV/AIDS in the Asian

- and Pacific Region —Ajay Tandon, November 2005 No. 76 Policy Reform in Indonesia and the Asian
- Development Bank's Financial Sector Governance Reforms Program Loan —George Abonyi, December 2005
- No. 77 Dynamics of Manufacturing Competitiveness in South Asia: ANalysis through Export Data —Hans-Peter Brunner and Massimiliano Calì, December 2005
- No. 78 Trade Facilitation —Teruo Ujiie, January 2006
- No. 79 An Assessment of Cross-country Fiscal Consolidation —Bruno Carrasco and Seung Mo Choi, February 2006

ERD TECHNICAL NOTE SERIES (TNS)

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

- No. 1 Contingency Calculations for Environmental Impacts with Unknown Monetary Values —David Dole, February 2002
 No. 2 Integrating Risk into ADB's Economic Analysis of Projects
 - —Nigel Rayner, Anneli Lagman-Martin, and Keith Ward, June 2002
- No. 3 Measuring Willingness to Pay for Electricity —Peter Choynowski, July 2002
- No. 4 Economic Issues in the Design and Analysis of a Wastewater Treatment Project —David Dole, July 2002
- No. 5 An Analysis and Case Study of the Role of Environmental Economics at the Asian Development Bank —David Dole and Piya Abeygunawardena, September 2002
- No. 6 Economic Analysis of Health Projects: A Case Study in Cambodia —Erik Bloom and Peter Chovnowski, May 2003
- No. 7 Strengthening the Economic Analysis of Natural Resource Management Projects —Keith Ward, September 2003
- No. 8 Testing Savings Product Innovations Using an Experimental Methodology

-Nava Ashraf, Dean S. Karlan, and Wesley Yin, November 2003

- No. 9 Setting User Charges for Public Services: Policies and Practice at the Asian Development Bank —David Dole, December 2003
- No. 10 Beyond Cost Recovery: Setting User Charges for Financial, Economic, and Social Goals —David Dole and Ian Bartlett, January 2004
- No. 11 Shadow Exchange Rates for Project Economic Analysis: Toward Improving Practice at the Asian Development Bank —Anneli Lagman-Martin, February 2004
- No. 12 Improving the Relevance and Feasibility of Agriculture and Rural Development Operational Designs: How Economic Analyses Can Help —Richard Bolt, September 2005
- No. 13 Assessing the Use of Project Distribution and Poverty Impact Analyses at the Asian Development Bank —Franklin D. De Guzman, October 2005
- No. 14 Assessing Aid for a Sector Development Plan: Economic Analysis of a Sector Loan
- -David Dole, November 2005 No. 15 Debt Management Analysis of Nepal's Public Debt -Sungsup Ra, Changyong Rhee, and Joon-Ho Hahm, December 2005

ERD POLICY BRIEF SERIES (PBS)

(Published in-house; Available through ADB Office of External Relations; Free of charge)

No. 1	Is Growth Good Enough for the Poor?
No 2	-Ernesto M. Pernia, October 2001 India's Economic Reforms
110. 2	What Has Been Accomplished?
	What Remains to Be Done?
	—Arvind Panagariya, November 2001
No. 3	Unequal Benefits of Growth in Viet Nam
	—Indu Bhushan, Erik Bloom, and Nguyen Minh
No 4	Inang, January 2002 Is Volatility Built into Today's World Economy?
110. 1	-J. Malcolm Dowling and J.P. Verbiest.
	February 2002
No. 5	What Else Besides Growth Matters to Poverty
	Reduction? Philippines
	-Arsenio M. Balisacan and Ernesto M. Pernia,
No 6	Achieving the Twin Objectives of Efficiency and
10. 0	Equity: Contracting Health Services in Cambodia
	—Indu Bhushan, Shervl Keller, and Brad Schwartz.
	March 2002
No. 7	Causes of the 1997 Asian Financial Crisis: What
	Can an Early Warning System Model Tell Us?
	—Juzhong Zhuang and Malcolm Dowling,
N. 0	June 2002
INO. 8	in Asia
	-Christopher Edmonds and Jean-Pierre Verbiest.
	July 2002
No. 9	The Doha Round: A Development Perspective
	—Jean-Pierre Verbiest, Jeffrey Liang, and Lea
N. 10	Sumulong, July 2002
NO. 10	Is Economic Openness Good for Regional
	Philippines
	-E. M. Pernia and Pilipinas Quising, October
	2002
No. 11	Implications of a US Dollar Depreciation for Asian
	Developing Countries
No 19	-Emma Fan, July 2002
110. 12	-D Brooks and Pilipinas Quising December 2002
No. 13	Infrastructure and Poverty Reduction—
	What is the Connection?
	—Ifzal Ali and Ernesto Pernia, January 2003
No. 14	Infrastructure and Poverty Reduction—
	Making Markets Work for the Poor
No. 15	-Alanoin 100, May 2005 SARS: Economic Impacts and Implications
110. 10	-Emma Xiaogin Fan. May 2003
No. 16	Emerging Tax Issues: Implications of Globalization
	and Technology
	—Kanokpan Lao Araya, May 2003
No. 17	Pro-Poor Growth: What is It and Why is It
	Ernesto M. Pernia, May 2003
No. 18	Public–Private Partnership for Competitiveness
	—Jesus Felipe, June 2003
No. 19	Reviving Asian Economic Growth Requires Further
	Reforms
No. 90	-Ifzal Ali, June 2003
No. 20	Are We Counting the World's Poor Right?
	-M. G. Quibria, July 2003
No. 21	Trade and Poverty: What are the Connections?
	-Douglas H. Brooks, July 2003
No. 22	Adapting Education to the Global Economy
NL CO	-Olivier Dupriez, September 2003
No. 23	Avian Flu: An Economic Assessment for Selected
	Developing Countries in Asia

—Jean-Pierre Verbiest and Charissa Castillo, March 2004

- No. 25 Purchasing Power Parities and the International Comparison Program in a Globalized World —Bishnu Pant, March 2004
- No. 26 A Note on Dual/Multiple Exchange Rates —Emma Xiaoqin Fan, May 2004
- No. 27 Inclusive Growth for Sustainable Poverty Reduction in Developing Asia: The Enabling Role of Infrastructure Development —Ifzal Ali and Xianbin Yao, May 2004
- No. 28 Higher Oil Prices: Asian Perspectives and Implications for 2004-2005
- -Cyn-Young Park, June 2004 Accelerating Acriculture and Burel
- No. 29 Accelerating Agriculture and Rural Development for Inclusive Growth: Policy Implications for Developing Asia —Richard Bolt, July 2004
- No. 30 Living with Higher Interest Rates: Is Asia Ready? —Cyn-Young Park, August 2004
- No. 31 Reserve Accumulation, Sterilization, and Policy Dilemma —Akiko Terada-Hagiwara, October 2004
- No. 32 The Primacy of Reforms in the Emergence of People's Republic of China and India —Ifzal Ali and Emma Xiaoqin Fan, November 2004
- No. 33 Population Health and Foreign Direct Investment: Does Poor Health Signal Poor Government Effectiveness? —Ajay Tandon, January 2005
- No. 34 Financing Infrastructure Development: Asian Developing Countries Need to Tap Bond Markets More Rigorously —Yun-Hwan Kim, February 2005
- No. 35 Attaining Millennium Development Goals in Health: Isn't Economic Growth Enough? —Ajay Tandon, March 2005
- No. 36 Instilling Credit Culture in State-owned Banks— Experience from Lao PDR —Robert Boumphrey, Paul Dickie, and Samiuela Tukuafu, April 2005
- No. 37 Coping with Global Imbalances and Asian Currencies —Cyn-Young Park, May 2005
- No. 38 Asia's Long-term Growth and Integration: Reaching beyond Trade Policy Barriers
 —Douglas H. Brooks, David Roland-Holst, and Fan Zhai, September 2005
- No. 39 Competition Policy and Development —Douglas H. Brooks, October 2005
- No. 40 Highlighting Poverty as Vulnerability: The 2005 Earthquake in Pakistan —Rana Hasan and Ajay Tandon, October 2005
- No. 41 Conceptualizing and Measuring Poverty as Vulnerability: Does It Make a Difference?
- -Ajay Tandon and Rana Hasan, October 2005 No. 42 Potential Economic Impact of an Avian Flu
- Pandemic on Asia —Erik Bloom, Vincent de Wit, and Mary Jane Carangal-San Jose, November 2005
- No. 43 Creating Better and More Jobs in Indonesia: A Blueprint for Policy Action —Guntur Sugiyarto, December 2005

SPECIAL STUDIES, COMPLIMENTARY

(Available through ADB Office of External Relations)

- 1. Improving Domestic Resource Mobilization Through Financial Development: Overview September 1985
- 2. Improving Domestic Resource Mobilization Through Financial Development: Bangladesh July 1986
- 3. Improving Domestic Resource Mobilization Through Financial Development: Sri Lanka April 1987
- 4. Improving Domestic Resource Mobilization Through Financial Development: India December 1987
- Financing Public Sector Development Expenditure in Selected Countries: Overview January 1988
 Study of Selected Industries: A Brief Report
- April 1988
- 7. Financing Public Sector Development Expenditure in Selected Countries: Bangladesh *June 1988*
- 8. Financing Public Sector Development Expenditure in Selected Countries: India June 1988
- 9. Financing Public Sector Development Expenditure in Selected Countries: Indonesia June 1988
- 10. Financing Public Sector Development Expenditure in Selected Countries: Nepal June 1988
- 11. Financing Public Sector Development Expenditure in Selected Countries: Pakistan June 1988
- 12. Financing Public Sector Development Expenditure in Selected Countries: Philippines June 1988
- 13. Financing Public Sector Development Expenditure in Selected Countries: Thailand June 1988
- Towards Regional Cooperation in South Asia: ADB/EWC Symposium on Regional Cooperation in South Asia February 1988
- 15. Evaluating Rice Market Intervention Policies: Some Asian Examples *April 1988*
- 16. Improving Domestic Resource Mobilization Through Financial Development: Nepal November 1988
- 17. Foreign Trade Barriers and Export Growth September 1988
- 18. The Role of Small and Medium-Scale Industries in the Industrial Development of the Philippines April 1989

- The Role of Small and Medium-Scale Manufacturing Industries in Industrial Development: The Experience of Selected Asian Countries January 1990
- 20. National Accounts of Vanuatu, 1983-1987 January 1990
- 21. National Accounts of Western Samoa, 1984-1986 February 1990
- Human Resource Policy and Economic Development: Selected Country Studies July 1990
- 23. Export Finance: Some Asian Examples September 1990
- 24. National Accounts of the Cook Islands, 1982-1986 September 1990
- 25. Framework for the Economic and Financial Appraisal of Urban Development Sector Projects January 1994
- 26. Framework and Criteria for the Appraisal and Socioeconomic Justification of Education Projects January 1994
- 27. Investing in Asia 1997 (Co-published with OECD)
- 28. The Future of Asia in the World Economy 1998 (Copublished with OECD)
- Financial Liberalisation in Asia: Analysis and Prospects 1999 (Co-published with OECD)
- Sustainable Recovery in Asia: Mobilizing Resources for Development 2000 (Co-published with OECD)
- 31. Technology and Poverty Reduction in Asia and the Pacific 2001 (Co-published with OECD)
- 32. Asia and Europe 2002 (Co-published with OECD)
- 33. Economic Analysis: Retrospective 2003
- 34. Economic Analysis: Retrospective: 2003 Update 2004
- 35. Development Indicators Reference Manual: Concepts and Definitions 2004
- 35. Investment Climate and Productivity Studies Philippines: Moving Toward a Better Investment Climate 2005

The Road to Recovery: Improving the Investment Climate in Indonesia $2005\,$

Sri Lanka: Improving the Rural and Urban Investment Climate 2005

OLD MONOGRAPH SERIES

(Available through ADB Office of External Relations; Free of charge)

EDRC REPORT SERIES (ER)

No. 1	ASEAN and the Asian Development Bank
	—Seiji Naya, April 1982
No. 2	Development Issues for the Developing East
	and Southeast Asian Countries
	and International Cooperation
	—Seiji Naya and Graham Abbott, April 1982
No. 3	Aid, Savings, and Growth in the Asian Region
	-J. Malcolm Dowling and Ulrich Hiemenz,
	April 1982
No. 4	Development-oriented Foreign Investment
	and the Role of ADB
	—Kiyoshi Kojima, April 1982
No. 5	The Multilateral Development Banks
	and the International Economy's Missing
	Public Sector
	—John Lewis, June 1982
No. 6	Notes on External Debt of DMCs
	—Evelyn Go, July 1982
No. 7	Grant Element in Bank Loans
	—Dal Hyun Kim, July 1982
No. 8	Shadow Exchange Rates and Standard
	Conversion Factors in Project Evaluation

 —Peter Warr, September 1982
 No. 9 Small and Medium-Scale Manufacturing Establishments in ASEAN Countries: Perspectives and Policy Issues
 —Mathias Bruch and Ulrich Hiemenz, January 1983

- No. 10 A Note on the Third Ministerial Meeting of GATT —Jungsoo Lee, January 1983
- No. 11 Macroeconomic Forecasts for the Republic of China, Hong Kong, and Republic of Korea —J.M. Dowling, January 1983
- No. 12 ASEAN: Economic Situation and Prospects —Seiji Naya, March 1983
- No. 13 The Future Prospects for the Developing Countries of Asia —Seiji Naya, March 1983
- No. 14 Energy and Structural Change in the Asia-Pacific Region, Summary of the Thirteenth Pacific Trade and Development Conference —Seiji Naya, March 1983
- No. 15 A Survey of Empirical Studies on Demand for Electricity with Special Emphasis on Price

Elasticity of Demand —Wisarn Pupphavesa, June 1983 Determinants of Paddy Production in Indonesia: No. 16 1972-1981-A Simultaneous Equation Model Approach —T.K. Jayaraman, June 1983 No. 17 The Philippine Economy: Economic Forecasts for 1983 and 1984 -J.M. Dowling, E. Go, and C.N. Castillo, June 1983 No. 18 Economic Forecast for Indonesia -J.M. Dowling, H.Y. Kim, Y.K. Wang, and C.N. Castillo, June 1983 No. 19 Relative External Debt Situation of Asian Developing Countries: An Application of Ranking Method -Jungsoo Lee, June 1983 New Evidence on Yields, Fertilizer Application, No. 20 and Prices in Asian Rice Production —William James and Teresita Ramirez, July 1983 No. 21 Inflationary Effects of Exchange Rate Changes in Nine Asian LDCs -Pradumna B. Rana and J. Malcolm Dowling, Jr., December 1983 Effects of External Shocks on the Balance No. 22 of Payments, Policy Responses, and Debt Problems of Asian Developing Countries —Seiji Naya, December 1983 No. 23 Changing Trade Patterns and Policy Issues: The Prospects for East and Southeast Asian **Developing Countries** -Seiji Naya and Ulrich Hiemenz, February 1984 No. 24 Small-Scale Industries in Asian Economic Development: Problems and Prospects —Seiji Naya, February 1984 No. 25 A Study on the External Debt Indicators Applying Logit Analysis -Jungsoo Lee and Clarita Barretto, February 1984 Alternatives to Institutional Credit Programs No. 26 in the Agricultural Sector of Low-Income Countries –Jennifer Sour. March 1984 Economic Scene in Asia and Its Special Features No. 27 -Kedar N. Kohli, November 1984 The Effect of Terms of Trade Changes on the No. 28 Balance of Payments and Real National Income of Asian Developing Countries -Jungsoo Lee and Lutgarda Labios, January 1985 No. 29 Cause and Effect in the World Sugar Market: Some Empirical Findings 1951-1982 —Yoshihiro Iwasaki, February 1985 Sources of Balance of Payments Problem No. 30 in the 1970s: The Asian Experience -Pradumna Rana, February 1985 India's Manufactured Exports: An Analysis No. 31 of Supply Sectors —Ifzal Ali, February 1985 No. 32 Meeting Basic Human Needs in Asian **Developing Countries** -Jungsoo Lee and Emma Banaria, March 1985 No. 33 The Impact of Foreign Capital Inflow on Investment and Economic Growth in Developing Asia -Evelyn Go, May 1985 The Climate for Energy Development No. 34 in the Pacific and Asian Region: Priorities and Perspectives –V.V. Desai, April 1986 Impact of Appreciation of the Yen on No. 35 Developing Member Countries of the Bank -Jungsoo Lee, Pradumna Rana, and Ifzal Ali, May 1986 No. 36 Smuggling and Domestic Economic Policies in Developing Countries

-A.H.M.N. Chowdhury, October 1986 No. 37 Public Investment Criteria: Economic Internal Rate of Return and Equalizing Discount Rate —Ifzal Ali, November 1986 No. 38 Review of the Theory of Neoclassical Political Economy: An Application to Trade Policies —M.G. Quibria, December 1986 No. 39 Factors Influencing the Choice of Location: Local and Foreign Firms in the Philippines -E.M. Pernia and A.N. Herrin, February 1987 No. 40 A Demographic Perspective on Developing Asia and Its Relevance to the Bank —E.M. Pernia, May 1987 No. 41 Emerging Issues in Asia and Social Cost Benefit Analysis —I. Ali, September 1988 Shifting Revealed Comparative Advantage: No. 42 Experiences of Asian and Pacific Developing Countries -P.B. Rana, November 1988 No. 43 Agricultural Price Policy in Asia: Issues and Areas of Reforms -I. Ali. November 1988 Service Trade and Asian Developing Economies No. 44 -M.G. Quibria. October 1989 No. 45 A Review of the Economic Analysis of Power Projects in Asia and Identification of Areas of Improvement —I. Ali, November 1989 Growth Perspective and Challenges for Asia: No. 46 Areas for Policy Review and Research -I. Ali, November 1989 No. 47 An Approach to Estimating the Poverty Alleviation Impact of an Agricultural Project —I. Ali, January 1990 No. 48 Economic Growth Performance of Indonesia, the Philippines, and Thailand: The Human Resource Dimension -E.M. Pernia, January 1990 Foreign Exchange and Fiscal Impact of a Project: No. 49 A Methodological Framework for Estimation -I. Ali. February 1990 Public Investment Criteria: Financial No. 50 and Economic Internal Rates of Return —I. Ali, April 1990 No. 51 Evaluation of Water Supply Projects: An Economic Framework —Arlene M. Tadle, June 1990 No. 52 Interrelationship Between Shadow Prices, Project

- Investment, and Policy Reforms: An Analytical Framework —I. Ali, November 1990 No. 53 Issues in Assessing the Impact of Project and Sector Adjustment Lending
- —I. Ali, December 1990
 No. 54 Some Aspects of Urbanization and the Environment in Southeast Asia
 —Ernesto M. Pernia, January 1991
- No. 55 Financial Sector and Economic Development: A Survey —Jungsoo Lee, September 1991
- No. 56 A Framework for Justifying Bank-Assisted Education Projects in Asia: A Review of the Socioeconomic Analysis and Identification of Areas of Improvement —Etienne Van De Walle, February 1992
- No. 57 Medium-term Growth-Stabilization Relationship in Asian Developing Countries and Some Policy Considerations —Yun-Hwan Kim, February 1993
- No. 58 Urbanization, Population Distribution, and Economic Development in Asia —Ernesto M. Pernia, February 1993
- No. 59 The Need for Fiscal Consolidation in Nepal:

The Results of a Simulation

-ruppo	aı	mauro	ana	Ronaia	Antonio	Buttong,
July 1993	3					

- No. 60 A Computable General Equilibrium Model of Nepal —*Timothy Buehrer and Filippo di Mauro, October* 1993
 No. 61 The Role of Government in Export Expansion
- in the Republic of Korea: A Revisit —Yun-Hwan Kim, February 1994 No. 62 Rural Reforms, Structural Change, and Agricultural Growth in the People's Republic of China

-Bo Lin, August 1994

- No. 63 Incentives and Regulation for Pollution Abatement with an Application to Waste Water Treatment -Sudipto Mundle, U. Shankar, and Shekhar Mehta, October 1995
- No. 64 Saving Transitions in Southeast Asia —Frank Harrigan, February 1996
- No. 65 Total Factor Productivity Growth in East Asia: A Critical Survey —Jesus Felipe, September 1997
- No. 66 Foreign Direct Investment in Pakistan:
 Policy Issues and Operational Implications
 —Ashfaque H. Khan and Yun-Hwan Kim, July 1999
- No. 67 Fiscal Policy, Income Distribution and Growth —Sailesh K. Jha, November 1999

ECONOMIC STAFF PAPERS (ES)

No.	1	International Reserves: Factors Determining Needs and Adequacy
NT.	0	-Evelyn Go, May 1981
INO.	Z	Agion Countries
		Asian Countries Basil Moore, assisted by A H M. Nuruddin
		-Busil Moore, assisted by A.H.M. Nuruaun Choudhum, Sontombor 1981
No	3	Changes in Consumption Imports and Exports
110.	0	of Oil Since 1973: A Preliminary Survey of
		the Developing Member Countries
		of the Asian Development Bank
		-Dal Hyun Kim and Graham Abbott, September
		1981
No.	4	By-Passed Areas, Regional Inequalities,
		and Development Policies in Selected
		Southeast Asian Countries
		—William James, October 1981
No.	5	Asian Agriculture and Economic Development
	_	—William James, March 1982
No.	6	Inflation in Developing Member Countries:
		An Analysis of Recent Trends
		-A.H.M. Nuruddin Chowdhury and J. Malcolm
No	7	Dowling, March 1982 Industrial Crowth and Employment in
110.	1	Doveloping Asian Countries: Issues and
		Perspectives for the Coming Decade
		—Illrich Hiemenz March 1982
No.	8	Petrodollar Recycling 1973-1980.
		Part 1: Regional Adjustments and
		the World Economy
		—Burnham Campbell, April 1982
No.	9	Developing Asia: The Importance
		of Domestic Policies
		—Economics Office Staff under the direction of Seiji
		Naya, May 1982
No.	10	Financial Development and Household
		Savings: Issues in Domestic Resource
		Wan Soon Kim July 1082
No	11	-Wan-Soon Kim, July 1982 Industrial Development: Role of Specialized
140.	11	Financial Institutions
		-Kedar N Kohli August 1982
No	12	Petrodollar Recycling 1973-1980
1.01		Part II: Debt Problems and an Evaluation
		of Suggested Remedies
		—Burnham Campbell, September 1982
No.	13	Credit Rationing, Rural Savings, and Financial
		Policy in Developing Countries
		—William James, September 1982
No.	14	Small and Medium-Scale Manufacturing
		Establishments in ASEAN Countries:
		Perspectives and Policy Issues
		<i>—Mathias Bruch and Ulrich Hiemenz, March 1983</i>

No. 15 Income Distribution and Economic

	Growth in Developing Asian Countries —J. Malcolm Dowling and David Soo. March 1983
No. 16	Long-Run Debt-Servicing Capacity of
	Asian Developing Countries: An Application
	of Critical Interest Rate Approach
No. 17	-Jungsoo Lee, June 1983 External Shocks, Energy Policy
100. 17	and Macroeconomic Performance of Asian
	Developing Countries: A Policy Analysis
	-William James, July 1983
No. 18	The Impact of the Current Exchange Rate
	System on Trade and Inflation of Selected
	Developing Member Countries
No. 10	-Pradumna Rana, September 1983
10. 19	-William James September 1983
No. 20	The Transition to an Industrial Economy
	in Monsoon Asia
	—Harry T. Oshima, October 1983
No. 21	The Significance of Off-Farm Employment
	and Incomes in Post-War East Asian Growth
N 00	-Harry T. Oshima, January 1984
No. 22	Asian Countries
	-John Malcolm Dowling Jr November 1984
No. 23	ASEAN Economies and ASEAN Economic
	Cooperation
	—Narongchai Akrasanee, November 1984
No. 24	Economic Analysis of Power Projects
N 05	-Nitin Desai, January 1985
No. 25	Exports and Economic Growth in the Asian Region
No. 26	Patterns of External Financing of DMCs
110. 20	-E. Go. May 1985
No. 27	Industrial Technology Development
	the Republic of Korea
	—S.Y. Lo, July 1985
No. 28	Risk Analysis and Project Selection:
	A Review of Practical Issues
No. 20	-J.K. Johnson, August 1985 Pice in Indenesia: Price Policy and Componenting
10. 29	Advantage
	-I. Ali, January 1986
No. 30	Effects of Foreign Capital Inflows
	on Developing Countries of Asia
	—Jungsoo Lee, Pradumna B. Rana, and Yoshihiro
	Iwasaki, April 1986
No. 31	Economic Analysis of the Environmental
	Impacts of Development Projects
	-John A. Dixon et al., EAFI, East-west Center, August 1986
No. 32	Science and Technology for Development:

No.	33	Satellite Remote Sensing in the Asian and Pacific Region
		—Mohan Sundara Rajan, December 1986
No.	34	Changes in the Export Patterns of Asian and
		Pacific Developing Countries: An Empirical
		Overview
		—Pradumna B. Rana, January 1987
No.	35	Agricultural Price Policy in Nepal
		-Gerald C. Nelson, March 1987
No	36	Implications of Falling Primary Commodity
		Prices for Agricultural Strategy in the Philippines
		—Ifzal Ali, Sentember 1987
No	37	Determining Irrigation Charges: A Framework
110.	0.	Prahlakar B. Ghate October 1987
No	38	The Bole of Fertilizer Subsidies in Agricultural
110.	00	Production: A Review of Select Issues
		-M.G. Quibria October 1987
No	30	Domestic Adjustment to External Shocks
110.	00	in Developing Asia
		Jungano Lag October 1087
No	40	Improving Demostic Resource Mobilization
110.	40	through Financial Development: Indenesia
		Dilin Enguiaga November 1097
No	11	-Fillip Erquiaga, November 1987
110.	41	Investment in Asian and Pasifia Developing
		Countries
		DD Drug March 1000
No	19	-F.B. Runu, Murch 1988 Manufactured Exports from the Philippines
110.	42	A Sector Profile and an Agenda for Peferm
		<i>L</i> Ali: Contember 1089
No	19	-1. All, September 1988
110.	40	A Framework for Evaluating the Economic
		L Ali August 1000
NT.	4.4	-I. All, August 1989
INO.	44	Promotion of Manufactured Exports in Pakistan
		-Jungsoo Lee ana Iosniniro Iwasaki, September
ЪT	15	1989
NO.	45	Education and Labor Markets in Indonesia:
		A Sector Survey
		-Ernesto M. Pernia and David N. Wilson,
		September 1989
No.	46	Industrial Technology Capabilities
		and Policies in Selected ADCs
	. –	—Hıroshi Kakazu, June 1990
No.	47	Designing Strategies and Policies
		for Managing Structural Change in Asia
		—Itzal Ali, June 1990

- No. 48 The Completion of the Single European Community Market in 1992: A Tentative Assessment of its Impact on Asian Developing Countries —J.P. Verbiest and Min Tang, June 1991
- No. 49 Economic Analysis of Investment in Power Systems —Ifzal Ali, June 1991
- No. 50 External Finance and the Role of Multilateral Financial Institutions in South Asia: Changing Patterns, Prospects, and Challenges —Jungsoo Lee, November 1991
- No. 51 The Gender and Poverty Nexus: Issues and Policies
- -M.G. Quibria, November 1993 No. 52 The Role of the State in Economic Development: Theory, the East Asian Experience, and the Malaysian Case -Jason Brown, December 1993
- No. 53 The Economic Benefits of Potable Water Supply Projects to Households in Developing Countries —Dale Whittington and Venkateswarlu Swarna, January 1994
- No. 54 Growth Triangles: Conceptual Issues and Operational Problems —Min Tang and Myo Thant, February 1994
- No. 55 The Emerging Global Trading Environment and Developing Asia —Arvind Panagariya, M.G. Quibria, and Narhari Rao, July 1996
- No. 56 Aspects of Urban Water and Sanitation in the Context of Rapid Urbanization in Developing Asia
 —Ernesto M. Pernia and Stella LF. Alabastro, September 1997
- No. 57 Challenges for Asia's Trade and Environment -Douglas H. Brooks, January 1998
- No. 58 Economic Analysis of Health Sector Projects-A Review of Issues, Methods, and Approaches —Ramesh Adhikari, Paul Gertler, and Anneli Lagman, March 1999
- No. 59 The Asian Crisis: An Alternate View —Rajiv Kumar and Bibek Debroy, July 1999
- No. 60 Social Consequences of the Financial Crisis in Asia —James C. Knowles, Ernesto M. Pernia, and Mary Racelis, November 1999

OCCASIONAL PAPERS (OP)

No. 1	Poverty in the People's Republic of China:
	Recent Developments and Scope
	for Bank Assistance
	—K.H. Moinuddin, November 1992
No. 2	The Eastern Islands of Indonesia: An Overview
	of Development Needs and Potential
	—Brien K. Parkinson, January 1993
No. 3	Rural Institutional Finance in Bangladesh
	and Nepal: Review and Agenda for Reforms
	—A.H.M.N. Chowdhury and Marcelia C. Garcia,
	November 1993
No. 4	Fiscal Deficits and Current Account Imbalances
	of the South Pacific Countries:
	A Case Study of Vanuatu
	—T.K. Jayaraman, December 1993
No. 5	Reforms in the Transitional Economies of Asia
	—Pradumna B. Rana, December 1993
No. 6	Environmental Challenges in the People's Republic
	of China and Scope for Bank Assistance
	—Elisabetta Capannelli and Omkar L. Shrestha,
	December 1993
No. 7	Sustainable Development Environment

and Poverty Nexus

	—K.F. Jalal, December 1993
No. 8	Intermediate Services and Economic
	Development: The Malaysian Example
	-Sutanu Behuria and Rahul Khullar, May 1994
No. 9	Interest Rate Deregulation: A Brief Survey
	of the Policy Issues and the Asian Experience
	-Carlos J. Glower, July 1994
No. 10	Some Aspects of Land Administration
	in Indonesia: Implications for Bank Operations
	—Sutanu Behuria, July 1994
No. 11	Demographic and Socioeconomic Determinants
	of Contraceptive Use among Urban Women in
	the Melanesian Countries in the South Pacific:
	A Case Study of Port Vila Town in Vanuatu
	—T.K. Jayaraman, February 1995
No. 12	Managing Development through
	Institution Building
	— Hilton L. Root, October 1995
No. 13	Growth, Structural Change, and Optimal
	Poverty Interventions
	—Shiladitya Chatterjee, November 1995

No. 14 Private Investment and Macroeconomic Environment in the South Pacific Island Countries: A Cross-Country Analysis

	—1.K. Jayaraman, October 1996
No. 15	The Rural-Urban Transition in Viet Nam:
	Some Selected Issues
	—Sudipto Mundle and Brian Van Arkadie, October
	1997
No. 16	A New Approach to Setting the Future
	Transport Agenda
	-Roger Allport, Geoff Key, and Charles Melhuish,
	June 1998
No. 17	Adjustment and Distribution:
	The Indian Experience
	-Sudipto Mundle and V.B. Tulasidhar, June 1998

No. 18	Tax Reforms in Viet Nam: A Selective Analysis	
	—Sudipto Mundle, December 1998	

- No. 19 Surges and Volatility of Private Capital Flows to Asian Developing Countries: Implications for Multilateral Development Banks —Pradumna B. Rana, December 1998
- No. 20 The Millennium Round and the Asian Economies: An Introduction
- -Dilip K. Das, October 1999 No. 21 Occupational Segregation and the Gender Earnings Gap -Joseph E. Zveglich, Jr. and Yana van der Meulen Rodgers, December 1999
- No. 22 Information Technology: Next Locomotive of Growth? —Dilip K. Das, June 2000

Survey of the External Debt Situation in Asian Developing Countries, 1985

STATISTICAL REPORT SERIES (SR)

No. 9

No. 1	Estimates of the Total External Debt of
	the Developing Member Countries of ADB:
	1981-1983
	—I.P. David, September 1984
No. 2	Multivariate Statistical and Graphical
	Classification Techniques Applied
	to the Problem of Grouping Countries
	-I.P. David and D.S. Maligalig, March 1985
No. 3	Gross National Product (GNP) Measurement
	Issues in South Pacific Developing Member
	Countries of ADB
	—S.G. Tiwari, September 1985
No. 4	Estimates of Comparable Savings in Selected
	DMCs
	—Hananto Sigit, December 1985
No. 5	Keeping Sample Survey Design
	and Analysis Simple
	—I.P. David, December 1985
No. 6	External Debt Situation in Asian
	Developing Countries
	-I.P. David and Jungsoo Lee, March 1986
No. 7	Study of GNP Measurement Issues in the
	South Pacific Developing Member Countries.
	Part I: Existing National Accounts
	of SPDMCs–Analysis of Methodology
	and Application of SNA Concepts
	-P. Hodgkinson, October 1986
No. 8	Study of GNP Measurement Issues in the South
	Pacific Developing Member Countries.
	Part II: Factors Affecting Intercountry
	Comparability of Per Capita GNP
	-P Hodgkinson October 1986

	—Jungsoo Lee and I.P. David, April 1987
No. 10	A Survey of the External Debt Situation
	in Asian Developing Countries, 1986
	—Jungsoo Lee and I.P. David, April 1988
No. 11	Changing Pattern of Financial Flows to Asian
	and Pacific Developing Countries
	–Jungsoo Lee and I.P. David, March 1989
No. 12	The State of Agricultural Statistics in
	Southeast Asia
	—I.P. David, March 1989
No. 13	A Survey of the External Debt Situation
	in Asian and Pacific Developing Countries:
	1987-1988
	—Jungsoo Lee and I.P. David, July 1989
No. 14	A Survey of the External Debt Situation in
	Asian and Pacific Developing Countries: 1988-1989
	—Jungsoo Lee, May 1990
No. 15	A Survey of the External Debt Situation
	in Asian and Pacific Developing Countries: 1989-
	1992
	—Min Tang, June 1991
No. 16	Recent Trends and Prospects of External Debt
	Situation and Financial Flows to Asian
	and Pacific Developing Countries
	—Min Tang and Aludia Pardo, June 1992
No. 17	Purchasing Power Parity in Asian Developing

- Countries: A Co-Integration Test —Min Tang and Ronald Q. Butiong, April 1994
- No. 18 Capital Flows to Asian and Pacific Developing Countries: Recent Trends and Future Prospects —Min Tang and James Villafuerte, October 1995

SERIALS

(Available commercially through ADB Office of External Relations)

- Asian Development Outlook (ADO; annual) \$36.00 (paperback)
- Key Indicators of Developing Asian and Pacific Countries (KI; annual) \$35.00 (paperback)
- 3. Asian Development Review (ADR; semiannual) \$5.00 per issue; \$10.00 per year (2 issues)

SPECIAL STUDIES, CO-PUBLISHED

(Available commercially through Oxford University Press Offices, Edward Elgar Publishing, and Palgrave MacMillan)

FROM OXFORD UNIVERSITY PRESS:

Oxford University Press (China) Ltd 18th Floor, Warwick House East Taikoo Place, 979 King's Road Quarry Bay, Hong Kong Tel (852) 2516 3222 Fax (852) 2565 8491 E-mail: webmaster@oupchina.com.hk Web: www.oupchina.com.hk

- Informal Finance: Some Findings from Asia Prabhu Ghate et. al., 1992 \$15.00 (paperback)
- Mongolia: A Centrally Planned Economy in Transition Asian Development Bank, 1992 \$15.00 (paperback)
- Rural Poverty in Asia, Priority Issues and Policy Options Edited by M.G. Quibria, 1994
- \$25.00 (paperback)Growth Triangles in Asia: A New Approach
- to Regional Economic Cooperation
 Edited by Myo Thant, Min Tang, and Hiroshi Kakazu
 1st ed., 1994 \$36.00 (hardbound)
 Revised ed., 1998 \$55.00 (hardbound)
- Urban Poverty in Asia: A Survey of Critical Issues Edited by Ernesto Pernia, 1994 \$18.00 (paperback)
- Critical Issues in Asian Development: Theories, Experiences, and Policies Edited by M.G. Quibria, 1995 \$15.00 (paperback) \$36.00 (hardbound)
- Financial Sector Development in Asia Edited by Shahid N. Zahid, 1995 \$50.00 (hardbound)
 Financial Sector Development in Asia: Country Studies
- Financial Sector Development in Asia: Country Studies Edited by Shahid N. Zahid, 1995 \$55.00 (hardbound)
- Fiscal Management and Economic Reform in the People's Republic of China Christine P.W. Wong, Christopher Heady, and Wing T. Woo, 1995 \$15.00 (paperback)
- 10. From Centrally Planned to Market Economies: The Asian Approach Edited by Pradumna B. Rana and Naved Hamid, 1995 Vol. 1: Overview \$36.00 (hardbound) Vol. 2: People's Republic of China and Mongolia \$50.00 (hardbound) Vol. 3: Lao PDR, Myanmar, and Viet Nam \$50.00 (hardbound)
 11. Current Issues in Economic Development:
- An Asian Perspective Edited by M.G. Quibria and J. Malcolm Dowling, 1996 \$50.00 (hardbound)
 The Bangladesh Economy in Transition Edited by M.G. Quibria, 1997
- \$20.00 (hardbound)13. The Global Trading System and Developing Asia
- Edited by Arvind Panagariya, M.G. Quibria, and Narhari Rao, 1997 \$55.00 (hardbound)
- 14. Social Sector Issues in Transitional Economies of Asia Edited by Douglas H. Brooks and Myo Thant, 1998

\$25.00 (paperback) \$55.00 (hardbound)

- Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Yun-Hwan Kim and Paul Smoke, 2003 \$15.00 (paperback)
- Local Government Finance and Bond Markets *Edited by Yun-Hwan Kim, 2003* \$15.00 (paperback)

FROM EDWARD ELGAR:

Marston Book Services Limited PO Box 269, Abingdon Oxon OX14 4YN, United Kingdom Tel +44 1235 465500 Fax +44 1235 465555 Email: direct.order@marston.co.uk Web: www.marston.co.uk

 Reducing Poverty in Asia: Emerging Issues in Growth, Targeting, and Measurement Edited by Christopher M. Edmonds, 2003

FROM PALGRAVE MACMILLAN:

Palgrave Macmillan Ltd Houndmills, Basingstoke Hampshire RG21 6XS, United Kingdom Tel: +44 (0)1256 329242 Fax: +44 (0)1256 479476 Email: orders@palgrave.com Web: www.palgrave.com/home/

- 1. Competition Policy and Development in Asia Edited by Douglas H. Brooks and Simon Evenett, 2005
- Managing FDI in a Globalizing Economy Asian Experiences Edited by Douglas H. Brooks and Hal Hill, 2004
- Poverty, Growth, and Institutions in Developing Asia Edited by Ernesto M. Pernia and Anil B. Deolalikar, 2003

SPECIAL STUDIES, IN-HOUSE

(Available commercially through ADB Office of External Relations)

- Rural Poverty in Developing Asia Edited by M.G. Quibria Vol. 1: Bangladesh, India, and Sri Lanka, 1994 \$35.00 (paperback) Vol. 2: Indonesia, Republic of Korea, Philippines, and Thailand, 1996 \$35.00 (paperback)
 Conden La Sictem of Developing Asian and Basif.
- Gender Indicators of Developing Asian and Pacific Countries Asian Development Bank, 1993 \$25.00 (paperback)
- External Shocks and Policy Adjustments: Lessons from the Gulf Crisis Edited by Naved Hamid and Shahid N. Zahid, 1995
- \$15.00 (paperback)4. Indonesia-Malaysia-Thailand Growth Triangle: Theory to Practice
- Edited by Myo Thant and Min Tang, 1996 \$15.00 (paperback)
- Emerging Asia: Changes and Challenges Asian Development Bank, 1997 \$30.00 (paperback)
- Asian Exports Edited by Dilip Das, 1999 \$35.00 (paperback) \$55.00 (hardbound)
- Development of Environment Statistics in Developing Asian and Pacific Countries Asian Development Bank, 1999 \$30.00 (paperback)
- Mortgage-Backed Securities Markets in Asia Edited by S.Ghon Rhee & Yutaka Shimomoto, 1999 \$35.00 (paperback)
- 9. Rising to the Challenge in Asia: A Study of Financial Markets
 - Asian Development Bank
 - Vol. 1: An Overview, 2000 \$20.00 (paperback)
 - Vol. 2: Special Issues, 1999 \$15.00 (paperback)
 - Vol. 3: Sound Practices, 2000 \$25.00 (paperback) Vol. 4: People's Republic of China, 1999 \$20.00
 - (paperback)
 - Vol. 5: India, 1999 \$30.00 (paperback)
 - Vol. 6: Indonesia, 1999 \$30.00 (paperback)
 - Vol. 7: Republic of Korea, 1999 \$30.00 (paperback)
 - Vol. 8: Malaysia, 1999 \$20.00 (paperback)
 - Vol. 9: Pakistan, 1999 \$30.00 (paperback)
 - Vol. 10: Philippines, 1999 \$30.00 (paperback)
 - Vol. 11: Thailand, *1999 \$30.00 (paperback)* Vol. 12: Socialist Republic of Viet Nam, *1999 \$30.00*
- (paperback)10. Corporate Governance and Finance in East Asia: A Study of Indonesia, Republic of Korea, Malaysia, Philippines and Thailand

J. Zhuang, David Edwards, D. Webb, & Ma. Virginita Capulong

Vol. 1: A Consolidated Report, 2000 \$10.00 (paperback)
Vol. 2: Country Studies, 2001 \$15.00 (paperback)
11. Financial Management and Governance Issues
Asian Development Bank, 2000
Cambodia \$10.00 (paperback)
People's Republic of China \$10.00 (paperback)

- Cambodia \$10.00 (paperback) People's Republic of China \$10.00 (paperback) Mongolia \$10.00 (paperback) Pakistan \$10.00 (paperback) Papua New Guinea \$10.00 (paperback) Uzbekistan \$10.00 (paperback) Viet Nam \$10.00 (paperback) Selected Developing Member Countries \$10.00 (paperback)
- Government Bond Market Development in Asia Edited by Yun-Hwan Kim, 2001
- \$25.00 (paperback)
 13. Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future *Edited by Paul Smoke and Yun-Hwan Kim, 2002*\$15.00 (paperback)
- Guidelines for the Economic Analysis of Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Guidelines for the Economic Analysis of Telecommunications Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Handbook for the Economic Analysis of Water Supply Projects Asian Development Bank, 1999 \$10.00 (hardbound)
- Handbook for the Economic Analysis of Health Sector Projects Asian Development Bank, 2000 \$10.00 (paperback)
- Handbook for Integrating Povery Impact Assessment in the Economic Analysis of Projects Asian Development Bank, 2001 \$10.00 (paperback)
- Handbook for Integrating Risk Analysis in the Economic Analysis of Projects Asian Development Bank, 2002 \$10.00 (paperback)
- Handbook on Environment Statistics Asian Development Bank, 2002 \$10.00 (hardback)
- Defining an Agenda for Poverty Reduction, Volume 1 Edited by Christopher Edmonds and Sara Medina, 2002 \$15.00 (paperback)
- Defining an Agenda for Poverty Reduction, Volume 2 Edited by Isabel Ortiz, 2002 \$15.00 (paperback)
- Economic Analysis of Policy-based Operations: Key Dimensions Asian Development Bank, 2003 \$10.00 (paperback)